

Available online at www.sciencedirect.com



Comput. Methods Appl. Mech. Engrg. 401 (2022) 115604

Computer methods in applied mechanics and engineering

www.elsevier.com/locate/cma

A computational framework for topology optimisation of flexoelectricity at finite strains considering a multi-field micromorphic approach

R. Ortigosa^a, J. Martínez-Frutos^a, A.J. Gil^{b,*}

^a Technical University of Cartagena, Campus Muralla del Mar, 30202, Cartagena (Murcia), Spain

^b Zienkiewicz Centre for Computational Engineering, Faculty of Science and Engineering, Swansea University, Bay Campus, SA1 8EN, United Kingdom

> Received 10 June 2022; received in revised form 22 August 2022; accepted 23 August 2022 Available online xxxx

Abstract

This paper presents a novel in-silico framework for the design of flexoelectric energy harvesters at finite strains using topology optimisation. The main ingredients of this work can be summarised as follows: (i) a micromorphic continuum approach is exploited to account for size dependent effects in the context of finite strains, thus permitting the modelling and simulation of flexoelectric effects in highly deformable materials such as dielectric elastomers. A key feature of the multi-field (mixed) formulation pursued is its flexibility as it permits, upon suitable selection of material parameters, to degenerate into other families of high order gradient theories such as flexoelectric gradient elasticity. (ii) A novel energy interpolation scheme is put forward, whereby different interpolation strategies are proposed for the various contributions that the free energy density function is decomposed into. This has enabled to circumvent numerical artifacts associated with fictitious high flexoelectric effects observed in the vicinity of low and intermediate density regions, where extremely high strain gradients tend to develop. (iii) A weighted combination of efficiency-based functionals, which promotes the development of hinges with unpractical highly localised large strain gradients. Finally, a series of numerical examples are analysed, studying the development of direct flexoelectricity induced by bending, compression and torsional deformations.

© 2022 The Author(s). Published by Elsevier B.V. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

Keywords: Flexoelectricity; Topology optimisation; Dielectric elastomer; Micromorphic elasticity; Energy harvesters; Mixed finite elements

1. Introduction

Flexoelectricity [1,2] is a two-way coupling mechanism between electric polarisation and inhomogeneous deformations (i.e., bending, torsion), only relevant at small length scales. To be more precise, *direct* flexoelectricity is responsible for the generation of polarisation in the material due to strain gradients. On the other hand, *converse* flexoelectricity accounts for the development of stresses within the material due to gradients of the electric field. Recent advancements in fabrication and design of miniaturised components have indeed motivated the interest of

* Corresponding author.

E-mail addresses: rogelio.ortigosa@upct.es (R. Ortigosa), jesus.martinez@upct.es (J. Martínez-Frutos), a.j.gil@swansea.ac.uk (A.J. Gil).

https://doi.org/10.1016/j.cma.2022.115604

^{0045-7825/© 2022} The Author(s). Published by Elsevier B.V. This is an open access article under the CC BY license (http://creativecommons. org/licenses/by/4.0/).

the scientific and engineering community for flexoelectric materials, leading to a vast number of publications in the experimental characterisation of this phenomenon [3–6].

Some of these works study the flexoelectric effect (at small length scales) inherent in piezoelectric crystals, such as Barium Titanate, devoid from a centrosymmetric internal structure. In these materials, it is interesting to quantify the electric energy originated either from piezoelectricity or flexoelectricity, when subjected to a given mechanical loading [5,7]. However, flexoelectricity is not exclusive from non-centrosymmetric materials, as shown in Refs. [8], demonstrating flexoelectricity in perovskite ferroelectrics [6], graphene nano-shells and biological membranes [3,4], all of them centrosymmetric. This is achievable when the inversion symmetry of these materials is broken through the application of non-uniform strains. Even dielectric elastomers, which are isotropic, and hence centrosymmetric, can exhibit flexoelectricity at small length scales [8]. The latter, namely dielectric elastomers, open up for new possibilities in the field of flexoelectric-based nano-actuators, sensors and energy harvesters, taking advantage of their capability of exhibiting large deformations [9-12]. At the macroscopic level, their ability to undergo large electrically induced deformations, in conjunction with their fast response, led to their identification as ideal candidates in the field of soft robotics. For instance, electrically induced area expansions of over 380% on dielectric elastomer thin films placed on the verge of snap-through configurations have been reported by Li et al. [13]. Using a theoretical framework, Ref. [14] demonstrates also the capability of dielectric elastomers balloons to undergo snap-through electrically induced actuations. Other applications for dielectric elastomers include Braille displays, deformable lenses, haptic devices and energy generators, to name but a few.

Flexoelectricity represents a generalisation of the simpler electro-elasticity continuum theory [15–18]. The variational formulation of the latter is extremely well established. In the most standard formulation, displacements and the scalar electric potential [15,19,20] are modelled as the unknown fields. In this formulation, the constitutive information is encapsulated in the Helmholtz energy functional via its invariant-based representation depending upon kinematic strain measures and the electric field [21–23]. However, for more complex constitutive models than that of an ideal dielectric elastomer, the saddle point nature of the Helmholtz functional (convex with respect to the deformation gradient tensor and concave with respect to the electric field in the small strain/small electric field regime), makes in general impossible to define a priori constitutive models which satisfy the ellipticity condition [24,25], which ensures the well-posedness of the problem. Motivated by the possible loss of ellipticity of the Helmholtz functional, Gil and Ortigosa [26–29] extended of the concept of polyconvexity [25,30–38] to the field of electro-mechanics by postulating a definition of the internal energy functional convex with respect to an extended set of arguments, proving that this definition satisfied the ellipticity condition unconditionally.

The previous classical continuum theory of electro-mechanics cannot be applied for the modelling of flexoelectricity. Instead, high order gradient theories and generalised continua [39] need to be used in order to account for size dependent effects inherent to flexoelectricity. For the particular case of dielectric elastomers, in the context of finite strains, there exist two approaches relying on two different types of high order continuum theories. On one hand, the works in [40–43] rely on continuum frameworks that lie within the classification of (flexoelectric) gradient elasticity. From the numerical point of view, the authors in [40,41] resort to different types of C^1 finite element approximations of the displacement field in order to handle the higher regularity entailed by the fourth order derivatives featuring in the formulation. However, the requirement of C^1 continuity in the displacement field approximation can be eliminated when mixed-type finite element formulations are employed. In this regard, see the work in [42,43], the latter using a couple stress-based approach for flexoelectricity. On the other hand, a more general approach for the modelling of flexoelectric materials is that pursued in [44], where a (flexoelectric) *micromorphic* approach in conjunction with a mixed-type finite element formulation is advocated for. The great advantage of the latter resides on its flexibility, in the sense that a suitable tuning of the material parameters featuring in the constitutive model permits to degenerate the formulation into that of flexoelectric gradient elasticity succinctly described previously.

Mathematical tools such as Topology Optimisation (TO) represent a powerful instrument with the potential to assist experienced researchers in the design of smart or stimuli responsive materials with complex underlying physical behaviours. The design of flexoelectric materials can indeed benefit from the use of such tools. TO of smart materials has been widely explored mainly in the context of linearised elasticity, making use of the wide spectrum of robust approaches available for TO, ranging from density-based methods, with the Solid Isotropic Material with Penalisation (SIMP) method as their maximum representative [45], level-set methods [46,47], phase-field methods [48], topological derivative methods [49] and evolutionary methods [50]. Kang and Wang [51] used

the SIMP method for the TO of piezoelectric ceramics, an electro-active material which unlike DEs, is restricted to scenarios characterised by very small deformations and/or displacements. Zhang et al. [52] also investigated the TO of piezoelectric sensors with active vibration control purposes. Other works [53–56] have investigated the simultaneous optimisation of polarisation and layout of piezoelectric ceramics over a fixed host (passive) material, and when the host structure is also included in the optimisation process [57–59]. Furthermore, the work in [60] carries out the TO of piezoelectric ceramic-based micro-grippers considering large displacements but small deformations. Relevant works in the field of TO of piezoelectric composites include [61-63]. Some recent work includes the design by means of TO of thermoelectric coolers and generators [64–66]. More recently, some works have ventured in the TO of dielectric elastomers. For instance, the work in Reference [67] delves into the TO of these materials with the aim of maximising the electrically induced rotation of a rotary device. Extremely interesting is the work in [68], where the TO is applied with the objective of conceiving intelligent microstructures for the design of wide tunable band gaps. Although not using a TO framework, the tunability of band gaps in the context of dielectric elastomers has also been investigated in a vast number of works (see for instance [69-71]) Furthermore, the recent works by Ortigosa et al. [72-74] presented a thorough numerical study on density-based and phase field-based TO in the context of nonlinear electro-mechanics [15–17,75,76], considering both DEs and piezoelectric polymers. That work proposes a novel energy interpolation scheme, where suggestions regarding the mechanical and electrical exponents featuring in the proposed interpolation scheme are made in order to yield designs devoid from undesired intermediate density regions.

Topology optimisation can help designing efficient energy harvesters taking advantage of the flexoelectric effect at small length scales. Few works exist on the effective application of TO techniques for the design of energy harvesters based on the principle of flexoelectricity [77,78]. The latter works are however restricted to the case of small strains. As shown in [77,78], embedding TO algorithms with the aim of maximising an efficiency-based objective function, typically measuring the ratio between the global (integral) mechanical energy and the global electric energy generated through flexoelectricity, leads to designs characterised by a poor structural performance, with hinges where high stress concentration occurs. This mechanical deficiency is indeed promoted by the inherent flexoelectric nature of the material, which promotes the development of hinges with highly localised large strain gradients. The latter leads to an extremely rapid growth of the electric energy generated when compared with the associated mechanical energy decay, yielding extremely efficient designs according to the metric chosen, but inappropriate from a structural/mechanical standpoint. In order to remedy the above shortcoming, this paper presents a new TO framework for the robust (safe) design of flexoelectric actuators with the following key novelties: (a) consideration of dielectric elastomers, hence, of finite strain scenarios; (b) definition of objective functions considering a weighted combination of efficiency-based measures and p-norm/KS aggregation/regularisation [79] functions of the Von Mises stress distribution within the material.

The layout of this paper is as follows: Section 2 introduces the flexoelectric micromorphic elasticity formulation presented in Ref. [44] in a self-contained manner. Section 3 describes all the ingredients of the optimisation problem by means of the SIMP method. Finally, Section 4 shows some numerical examples demonstrating the applicability of the proposed formulation. Finally, Section 5 provides some concluding remarks.

2. Flexoelectric micromorphic elasticity

An introduction into nonlinear continuum electromechanics considering a micromorphic approach will be covered in this section, as well as the relevant governing equations.

2.1. Kinematics in the macro-scale

Let us consider the motion of a continuum with reference configuration $\mathcal{B}_0 \subset \mathbb{R}^n$ $(n = \{2, 3\})$. After the motion, the solid occupies a deformed configuration $\mathcal{B} \subset \mathbb{R}^n$ (refer to Fig. 1). We assume the existence of a deformation mapping $\phi(X)$ linking material particles from the reference configuration $X \in \mathcal{B}_0$ to the deformed configuration $x \in \mathcal{B}$ according to $x = \phi(X)$. Associated with $\phi(X)$, the deformation gradient tensor F [80–83] is defined as

$$\boldsymbol{F} = \boldsymbol{\nabla}_0 \boldsymbol{\phi} \left(\boldsymbol{X} \right); \qquad F_{iI} = \frac{\partial \phi_i}{\partial X_I}, \tag{1}$$

which permits to define its co-factor and its Jacobian as

$$H = (\det F) F^{-T}; \qquad J = \det F.$$
(2)



Fig. 1. Deformation mapping $\phi(X, t)$. The DE in its material (\mathcal{B}_0) and deformed (\mathcal{B}) configurations. Microelement around point P (p in spatial configuration) both in the material and spatial configurations, namely $\mathcal{B}_0^{X_p}$ and \mathcal{B}^{X_p} , respectively.

2.2. The micromorphic problem

In this paper, a micromorphic approach is considered in order to mathematically describe the continuum \mathcal{B}_0 . Hence, attached to every macroscopic material point $X \in \mathcal{B}_0$, there exists a microelement \mathcal{B}_0^X , centred in X. Microscopic points in \mathcal{B}_0^X are denoted as $\Xi \in \mathcal{B}_0^X$. Similarly, attached to every macroscopic spatial point $x \in \mathcal{B}$, there exists a microelement \mathcal{B}^x , centred in x. Microscopic points in \mathcal{B}^x are denoted as $\xi \in \mathcal{B}^x$. The latter are related with their material counterparts $\Xi \in \mathcal{B}_0^X$ through the following affine transformation [39]

$$\boldsymbol{\xi} = \boldsymbol{G}\boldsymbol{\Xi},\tag{3}$$

where $G \in \mathbb{R}^{n \times n}$ represents the microscopic deformation gradient tensor. Notice that in the micromorphic continuum, both microscopic and macroscopic deformation gradient tensors are independent fields, i.e. $F \neq G$. Associated with G, it is possible to define its gradient with respect to the macroscale material coordinates $X \in \mathcal{B}_0$ as

$$\mathcal{G} = \nabla_0 G; \qquad \mathcal{G}_{iIJ} = \partial_{X_J} G_{iI}. \tag{4}$$

2.3. Governing equations

In the absence of inertia effects, the local form of the conservation of linear momentum [83] in the solid \mathcal{B}_0 can be written as

DIV
$$P + f_0 = 0;$$
 in $\mathcal{B}_0;$
 $PN = t_0;$ on $\partial_{t_0} \mathcal{B}_0;$
 $\phi = \bar{\phi};$ on $\partial_{\phi} \mathcal{B}_0,$
(5)

where f_0 represents a body force per unit undeformed volume \mathcal{B}_0 and t_0 , the traction force per unit undeformed area on $\partial_{t_0}\mathcal{B}_0$. Furthermore, $\bar{\phi}$ represents the value of the Dirichlet boundary condition on $\partial_{\phi}\mathcal{B}_0$, and with N the outward unit vector on the boundary, and $\partial_{t_0}\mathcal{B}_0 \cup \partial_{\phi}\mathcal{B}_0 = \partial \mathcal{B}_0$ and $\partial_{t_0}\mathcal{B}_0 \cap \partial_{\phi}\mathcal{B}_0 = \emptyset$. Finally, P represents the macroscopic first Piola–Kirchhoff stress tensor. With regard to the equations for electrostatics in the macroscale, these reduce to Gauss's and Faraday's laws on \mathcal{B}_0 . The local form of the Gauss' law [76,84] can be written in a Lagrangian formalism as

DIV
$$\boldsymbol{D}_0 - \rho_0 = 0;$$
 in $\mathcal{B}_0;$
 $\boldsymbol{D}_0 \cdot \boldsymbol{N} = -\omega_0;$ on $\partial_{\omega_0} \mathcal{B}_0;$
(6)

where D_0 is the Lagrangian electric displacement vector, ρ_0 represents an electric volume charge per unit of undeformed volume \mathcal{B}_0 and ω_0 , an electric surface charge per unit of undeformed area $\partial_{\omega_0} \mathcal{B}_0 \subset \partial \mathcal{B}_0$. Furthermore, under the same hypothesis, the local form of the Faraday's law can be written in a Lagrangian setting as

where E_0 is the Lagrangian electric field vector and φ , the scalar electric potential. In (7), $\partial_{\varphi}\mathcal{B}_0$ represents the part of the boundary $\partial \mathcal{B}_0$ where essential electric potential boundary conditions are applied such that $\partial_{\omega_0}\mathcal{B}_0 \cup \partial_{\varphi}\mathcal{B}_0 = \partial \mathcal{B}_0$ and $\partial_{\omega_0}\mathcal{B}_0 \cap \partial_{\varphi}\mathcal{B}_0 = \emptyset$. The system of coupled PDEs is completed with the governing equations associated with the micromorphic continuum, which are expressed in a Lagrangian setting as [44]

DIV
$$\Sigma_{\mathcal{G}} - \Sigma_{G} = \mathbf{0};$$
 in $\mathcal{B}_{0};$
 $\Sigma_{\mathcal{G}} N = -T_{0};$ on $\partial_{T_{0}} \mathcal{B};$
 $G = \bar{G};$ on $\partial_{G} \mathcal{B},$
(8)

where $\Sigma_{\mathcal{G}} \in \mathbb{R}^{3 \times 3 \times 3}$ and $\Sigma_{\mathcal{G}} \in \mathbb{R}^{3 \times 3}$ represent the micromorphic double stress and the micromorphic Piola stress, respectively. Furthermore, $T_0 \in \mathbb{R}^{3 \times 3}$ represents the micromorphic Piola traction on $\partial_{T_0} \mathcal{B}_0$ and $\bar{\mathcal{G}}$, the prescribed value of \mathcal{G} on $\partial_{\mathcal{G}} \mathcal{B}_0$, such that $\partial_{T_0} \mathcal{B}_0 \cup \partial_{\mathcal{G}} \mathcal{B}_0 = \partial \mathcal{B}_0$ and $\partial_{T_0} \mathcal{B}_0 \cap \partial_{\mathcal{G}} \mathcal{B}_0 = \emptyset$.

Remark 1. As indicated in Steinman et al. [44], the physical meaning of Dirichlet boundary conditions on the field G is not completely understood. Nonetheless, this type of boundary conditions have been included in (8) for the sake of completeness. However, in the numerical examples in Section 4, only natural (or Neumann) boundary conditions are considered for this field.

2.4. Variational formulation

In this section we will present the variational formulation associated with the nonlinear PDEs in Eqs. (5), (6), (7) and (8), which stems from a three-field variational principle with unknown fields $\mathcal{U} = \{\phi, \varphi, G\}$ belonging to suitable functional spaces

$$\mathbb{V}^{\boldsymbol{\phi}} = \left\{ \boldsymbol{\phi} : \mathcal{B}_{0} \to \mathbb{R}^{3}, \quad (\boldsymbol{\phi})_{i} \in H^{1}(\mathcal{B}_{0}) \right\}; \qquad \mathbb{V}^{\varphi} = \left\{ \varphi : \mathcal{B}_{0} \to \mathbb{R}, \qquad \varphi \in H^{1}(\mathcal{B}_{0}) \right\}; \\
\mathbb{V}^{\boldsymbol{G}} = \left\{ \boldsymbol{G} : \mathcal{B}_{0} \to \mathbb{R}^{3 \times 3}, \quad (\boldsymbol{G})_{iI} \in H^{1}(\mathcal{B}_{0}) \right\},$$
(9)

where $H^1(\mathcal{B}_0)$ stands for the Sobolev space of functions belonging to $L^2(\mathcal{B}_0)$ (square summable functions) and with gradient also in L^2 . The three-field variational principle, denoted as $\Pi(\mathcal{U})$, can be defined as

$$\Pi(\mathcal{U}) = \inf_{\phi, G} \sup_{\varphi} \left\{ \int_{\mathcal{B}_0} \Psi(\mathcal{F}) \, dV - \Pi^{\text{ext}}(\phi, \varphi, G) \right\}, \qquad \mathcal{F} = \{F, E_0, G, \mathcal{G}\}, \tag{10}$$

where $\Psi(\mathcal{F})$ represents the free energy density of the material. In above equation, it is implicitly assumed that $\{F, E_0, \mathcal{G}\}$ depend upon the gradients of the unknown fields $\mathcal{U} = \{\phi, \varphi, G\}$ through Eqs. (1), (7) and (4), respectively. In addition, the external contribution $\Pi^{\text{ext}}(\phi, \varphi, G)$ is decomposed into volumetric and surface contributions, i.e.

$$\Pi^{\text{ext}}(\boldsymbol{\phi},\varphi,\boldsymbol{G}) = \Pi^{\text{ext}}_{\mathcal{B}_0}(\boldsymbol{\phi},\varphi) + \Pi^{\text{ext}}_{\partial\mathcal{B}_0}(\boldsymbol{\phi},\varphi,\boldsymbol{G}), \tag{11}$$

with

$$\Pi_{\mathcal{B}_{0}}^{\text{ext}}(\boldsymbol{\phi},\varphi) = \int_{\mathcal{B}_{0}} \boldsymbol{f}_{0} \cdot \boldsymbol{\phi} \, dV - \int_{\mathcal{B}_{0}} \rho_{0} \varphi \, dV; \qquad \Pi_{\partial \mathcal{B}_{0}}^{\text{ext}}(\boldsymbol{\phi},\varphi) = \int_{\partial_{t_{0}} \mathcal{B}_{0}} \boldsymbol{t}_{0} \cdot \boldsymbol{\phi} \, dV - \int_{\partial_{\omega_{0}} \mathcal{B}_{0}} \omega_{0} \varphi \, dV + \int_{\partial_{T_{0}} \mathcal{B}_{0}} \boldsymbol{T}_{0} : \boldsymbol{G} \, dV.$$

$$(12)$$

The directional derivatives of $\Pi(\mathcal{U})$ in (10) with respect to $p_{\mathcal{U}} = \{p_{\phi}, p_{\phi}, p_{G}\} \in \mathbb{V}_{0}^{\phi} \times \mathbb{V}_{0}^{\phi} \times \mathbb{V}_{0}^{G}$ yield the weak forms of (5), (6) and (8), respectively, namely

$$D\Pi[\boldsymbol{p}_{\phi}] = \int_{\mathcal{B}_{0}} \partial_{\boldsymbol{F}} \Psi(\boldsymbol{\mathcal{F}}) : \boldsymbol{\nabla}_{0} \boldsymbol{p}_{\phi} \, dV - \int_{\mathcal{B}_{0}} \boldsymbol{f}_{0} \cdot \boldsymbol{p}_{\phi} - \int_{\partial_{\boldsymbol{t}_{0}} \mathcal{B}_{0}} \boldsymbol{t}_{0} \cdot \boldsymbol{p}_{\phi} \, dA = 0;$$

$$D\Pi[\boldsymbol{p}_{\varphi}] = -\int_{\mathcal{B}_{0}} \partial_{\boldsymbol{E}_{0}} \Psi(\boldsymbol{\mathcal{F}}) \cdot \boldsymbol{\nabla}_{0} \boldsymbol{p}_{\varphi} \, dV + \int_{\mathcal{B}_{0}} \rho_{0} \boldsymbol{p}_{\varphi} + \int_{\partial_{\omega_{0}} \mathcal{B}_{0}} \omega_{0} \boldsymbol{p}_{\varphi} \, dA = 0;$$

$$D\Pi[\boldsymbol{p}_{G}] = \int_{\mathcal{B}_{0}} \partial_{\boldsymbol{G}} \Psi(\boldsymbol{\mathcal{F}}) : \boldsymbol{p}_{G} \, dV + \int_{\mathcal{B}_{0}} \partial_{\boldsymbol{\mathcal{G}}} \Psi(\boldsymbol{\mathcal{F}}) : \boldsymbol{\nabla}_{0} \boldsymbol{p}_{G} \, dV - \int_{\partial_{\boldsymbol{T}_{0}} \mathcal{B}_{0}} \boldsymbol{T}_{0} : \boldsymbol{p}_{G} \, dA = 0.$$
(13)

Integration by parts an application of the Gauss' theorem on Eqs. (13), permits to conclude that these correspond with the weak forms of the boundary value problem associated with Eqs. (5), (6), (7) and (8), provided that $\{P, D_0, \Sigma_G, \Sigma_G\}$ are related with the derivatives of the free energy density $\Psi(F, E_0, G, \mathcal{G})$ featuring in (13) through the following relationships

$$\boldsymbol{P} = \partial_{\boldsymbol{F}} \Psi(\boldsymbol{\mathcal{F}}); \qquad \boldsymbol{D}_0 = -\partial_{\boldsymbol{E}_0} \Psi(\boldsymbol{\mathcal{F}}); \qquad \boldsymbol{\Sigma}_{\boldsymbol{G}} = \partial_{\boldsymbol{G}} \Psi(\boldsymbol{\mathcal{F}}); \qquad \boldsymbol{\Sigma}_{\boldsymbol{\mathcal{G}}} = \partial_{\boldsymbol{\mathcal{G}}} \Psi(\boldsymbol{\mathcal{F}}).$$
(14)

The stationary conditions in (A.3), namely $D\Pi[p_{\phi}] = 0$, $D\Pi[p_{\phi}] = 0$ and $D\Pi[p_G] = 0$, are coupled through the nonlinearity of the free energy density $\Psi(F, E_0, G, G)$. It is customary to make use of a Newton– Raphson scheme where the three weak forms can be linearised with respect to incremental fields $\{\Delta\phi, \Delta\varphi, \Delta G\} \in \mathbb{V}_0^{\phi^h} \times \mathbb{V}_0^{\phi^h} \times \mathbb{V}_0^{\phi^h}$ as

$$0 = D\Pi[p_{\phi}] + D\Pi[p_{\phi}] + D\Pi[p_{G}]$$

+ $D^{2}\Pi[p_{\phi}; \Delta\phi] + D^{2}\Pi[p_{\phi}; \Delta\phi] + D^{2}\Pi[p_{\phi}; \Delta G]$
+ $D^{2}\Pi[p_{\phi}; \Delta\phi] + D^{2}\Pi[p_{\phi}; \Delta\phi] + D^{2}\Pi[p_{\phi}; \Delta G]$
+ $D^{2}\Pi[p_{G}; \Delta\phi] + D^{2}\Pi[p_{G}; \Delta\phi] + D^{2}\Pi[p_{G}; \Delta G],$ (15)

which permit the update of $\{\phi, \varphi, G\} \in \mathbb{V}^{\phi^h} \times \mathbb{V}^{\varphi^h} \times \mathbb{V}^{G^h}$ at a given Newton–Raphson iteration k + 1 as

$$\boldsymbol{\phi}^{k+1} = \boldsymbol{\phi}^k + \Delta \boldsymbol{\phi}; \qquad \boldsymbol{\varphi}^{k+1} = \boldsymbol{\varphi}^k + \Delta \boldsymbol{\varphi}; \qquad \boldsymbol{G}^{k+1} = \boldsymbol{G}^k + \Delta \boldsymbol{G}.$$
(16)

In (15), the three terms $D^2 \Pi[p_{\phi}; \Delta \phi]$, $D^2 \Pi[p_{\phi}; \Delta \varphi]$ and $D^2 \Pi[p_{\phi}; \Delta G]$, stemming from the linearisation of $D \Pi[p_{\phi}]$ with respect to $\Delta \mathcal{U}$, are obtained as

$$D^{2}\Pi[\boldsymbol{p}_{\phi}; \Delta \boldsymbol{\phi}] = \int_{\mathcal{B}_{0}} \nabla_{0}\boldsymbol{p}_{\phi} : \partial_{FF}^{2} \Psi(\mathcal{F}) : \nabla_{0}\Delta \boldsymbol{\phi} \, dV;$$

$$D^{2}\Pi[\boldsymbol{p}_{\phi}; \Delta \varphi] = -\int_{\mathcal{B}_{0}} \nabla_{0}\boldsymbol{p}_{\phi} : \partial_{FE_{0}}^{2} \Psi(\mathcal{F}) \cdot \nabla_{0}\Delta \varphi \, dV;$$

$$D^{2}\Pi[\boldsymbol{p}_{\phi}; \Delta \boldsymbol{G}] = \int_{\mathcal{B}_{0}} \nabla_{0}\boldsymbol{p}_{\phi} : \partial_{FG}^{2} \Psi(\mathcal{F}) : \Delta \boldsymbol{G} \, dV + \int_{\mathcal{B}_{0}} \nabla_{0}\boldsymbol{p}_{\phi} : \partial_{FG}^{2} \Psi(\mathcal{F}) : \nabla_{0}\Delta \boldsymbol{G} \, dV$$
(17)

Furthermore, in (15), the three terms $D^2 \Pi[p_{\varphi}; \Delta \phi]$, $D^2 \Pi[p_{\varphi}; \Delta \phi]$ and $D^2 \Pi[p_{\varphi}; \Delta G]$, stemming from the linearisation of $D \Pi[p_{\varphi}]$ with respect to $\Delta \mathcal{U}$, are obtained as

$$D^{2}\Pi[p_{\varphi};\Delta\boldsymbol{\phi}] = -\int_{\mathcal{B}_{0}} \nabla_{0}p_{\varphi} \cdot \partial_{E_{0}F}\Psi(\mathcal{F}) : \nabla_{0}\Delta\boldsymbol{\phi}\,dV;$$

$$D^{2}\Pi[p_{\varphi};\Delta\varphi] = \int_{\mathcal{B}_{0}} \nabla_{0}p_{\varphi} \cdot \partial_{E_{0}E_{0}}^{2}\Psi(\mathcal{F})\nabla_{0}\Delta\varphi\,dV;$$

$$D^{2}\Pi[p_{\varphi};\Delta\boldsymbol{G}] = -\int_{\mathcal{B}_{0}} \nabla_{0}p_{\varphi} : \partial_{E_{0}G}^{2}\Psi(\mathcal{F}) : \Delta\boldsymbol{G}\,dV - \int_{\mathcal{B}_{0}} \nabla_{0}p_{\varphi} \cdot \partial_{E_{0}\mathcal{G}}^{2}\Psi(\mathcal{F}) : \nabla_{0}\Delta\boldsymbol{G}\,dV;$$
(18)

Finally, linearisation of $D\Pi[p_G]$ with respect to $\Delta \mathcal{U}$ yields

$$D^{2}\Pi[\boldsymbol{p}_{G}; \Delta\boldsymbol{\phi}] = \int_{\mathcal{B}_{0}} \boldsymbol{p}_{G} : \partial_{\boldsymbol{G}\boldsymbol{F}}^{2}\Psi(\boldsymbol{\mathcal{F}}) : \nabla_{0}\Delta\boldsymbol{\phi}\,dV + \int_{\mathcal{B}_{0}} \nabla_{0}\boldsymbol{p}_{G} : \partial_{\boldsymbol{G}}^{2}\boldsymbol{F}\,\Psi(\boldsymbol{\mathcal{F}}) : \nabla_{0}\Delta\boldsymbol{\phi}\,dV;$$

$$D^{2}\Pi[\boldsymbol{p}_{G}; \Delta\boldsymbol{\phi}] = -\int_{\mathcal{B}_{0}} \boldsymbol{p}_{G} : \partial_{\boldsymbol{G}\boldsymbol{E}_{0}}^{2}\Psi(\boldsymbol{\mathcal{F}}) \cdot \nabla_{0}\Delta\boldsymbol{\phi}\,dV - \int_{\mathcal{B}_{0}} \nabla_{0}\boldsymbol{p}_{G} : \partial_{\boldsymbol{G}}^{2}\boldsymbol{E}_{0}\Psi(\boldsymbol{\mathcal{F}}) \cdot \nabla_{0}\Delta\boldsymbol{\phi}\,dV;$$

$$D^{2}\Pi[\boldsymbol{p}_{G}; \Delta\boldsymbol{G}] = \int_{\mathcal{B}_{0}} \boldsymbol{p}_{G} : \partial_{\boldsymbol{G}\boldsymbol{G}}^{2}\Psi(\boldsymbol{\mathcal{F}}) : \Delta\boldsymbol{G}\,dV + \int_{\mathcal{B}_{0}} \boldsymbol{p}_{G} : \partial_{\boldsymbol{G}}^{2}\boldsymbol{\mathcal{G}}\,\Psi(\boldsymbol{\mathcal{F}}) : \nabla_{0}\Delta\boldsymbol{G}\,dV + \int_{\mathcal{B}_{0}} \nabla_{0}\boldsymbol{p}_{G} : \partial_{\boldsymbol{G}}^{2}\boldsymbol{\mathcal{G}}\,\Psi(\boldsymbol{\mathcal{F}}) : \nabla_{0}\Delta\boldsymbol{G}\,dV + \int_{\mathcal{B}_{0}} \nabla_{0}\boldsymbol{p}_{G} : \partial_{\boldsymbol{G}}^{2}\boldsymbol{\mathcal{G}}\,\Psi(\boldsymbol{\mathcal{F}}) : \nabla_{0}\Delta\boldsymbol{G}\,dV.$$

$$(19)$$

The details concerning the Finite Element discretisation of the nonlinear stationary conditions in (13) and their linearisations in (17)–(19) can be found in Appendix A.

2.5. Constitutive equations

The free energy density of the material $\Psi(\mathcal{F})$ (with $\mathcal{F} = \{F, E_0, G, \mathcal{G}\}$) was introduced in the three-field variational principle $\Pi(\mathcal{U})$ in Eq. (10), with

$$\Psi = \Psi(\mathcal{F}) : \mathbb{R}^{3 \times 3} \times \mathbb{R}^3 \times \mathbb{R}^{3 \times 3} \times \mathbb{R}^{3 \times 3 \times 3} \to \mathbb{R}.$$
(20)

As shown in Eq. (14), the derivatives of $\Psi(\mathcal{F})$ with respect to \mathcal{F} permit to obtain their respective work conjugates, namely $\{P, D_0, \Sigma_G, \Sigma_G\}$. The objective of this section is to specify the constitutive model followed in this paper, namely, the explicit expression for the free energy density $\Psi(\mathcal{F})$. In order to do so, we anticipate that the have followed the same model as that considered in Ref. [44], where the free energy density is additively decomposed into four contributions as

$$\Psi(\mathcal{F}) = \Psi^{\mathrm{mac}}(F, E_0) + \Psi^{\mathrm{mic}}(\mathcal{G}) + \Psi^{\mathrm{scale}}(F, G) + \Psi^{\mathrm{flexo}}(E_0, G, \mathcal{G}),$$
(21)

where $\Psi^{\text{mac}}(F, E_0)$ stands for the macroscopic contribution, $\Psi^{\text{mic}}(\mathcal{G})$ refers to the micromorphic contribution, $\Psi^{\text{scale}}(F, G)$ is the scale-bridging contribution and finally, $\Psi^{\text{flexo}}(E_0, G, \mathcal{G})$ is the flexoelectric contribution, being responsible for the flexoelectric effect. In the subsequent sections, each of the contributions of the free energy density will be specified.

2.5.1. The macroscopic free energy density

The macroscopic energetic contribution, namely $\Psi^{\text{mac}}(\mathbf{F}, \mathbf{E}_0)$, corresponds with the free energy density function of a dielectric elastomer, capable of exhibiting large electrically induced deformations. As customary in nonlinear electro-mechanics, the energy functional $\Psi^{\text{mac}}(\mathbf{F}, \mathbf{E}_0)$ is additively decomposed into a purely mechanical contribution and a coupled contribution, denoted as $\Psi^{\text{macro}}_{\text{m}}$ and $\Psi^{\text{macro}}_{\text{em}}$, respectively, as

$$\Psi^{\rm mac}(F, E_0) = \Psi^{\rm mac}_{\rm m}(F) + \Psi^{\rm mac}_{\rm em}(F, E_0).$$
(22)

For the mechanical contribution $\Psi_{\rm m}^{\rm mac}(F)$, polyconvex [25] free energy density function will be considered, namely any function that can be re-written as a convex multivariable function in terms of the extended set of kinematic entities $\mathcal{V} = \{F, H, J\}$, i.e.

$$\Psi_{\rm m}^{\rm mac}(\boldsymbol{F}) = \mathbb{W}_{\rm m}^{\rm mac}(\boldsymbol{\mathcal{V}}); \qquad \mathbb{W}_{\rm m}^{\rm mac}\left(\theta \boldsymbol{\mathcal{V}}_1 + (1-\theta) \boldsymbol{\mathcal{V}}_2\right) \le \theta \mathbb{W}_{\rm m}^{\rm mac}(\boldsymbol{\mathcal{V}}_1) + (1-\theta) \mathbb{W}_{\rm m}^{\rm mac}(\boldsymbol{\mathcal{V}}_2), \tag{23}$$

 $\forall \mathcal{V}_1, \mathcal{V}_2 \in \mathbb{R}^{3 \times 3} \times \mathbb{R}^{3 \times 3} \times \mathbb{R}^+$ and $\theta \in [0, 1]$. Specifically, in this paper we consider the Mooney–Rivlin free energy function to describe the macroscopic contribution of the free energy $\Psi(\mathcal{F})$, which is defined as

$$\Psi_{\rm m}^{\rm mac}(\boldsymbol{F}) = \mathbb{W}_{\rm m}^{\rm mac}(\boldsymbol{F}, \boldsymbol{H}, J) = \frac{\mu_1}{2} \left(||\boldsymbol{F}||^2 - 3 \right) + \frac{\mu_2}{2} \left(||\boldsymbol{H}||^2 - 3 \right) + f(J);$$

$$f(J) = -\left(\mu_1 + 2\mu_2\right) \ln J + \frac{\lambda}{2} \left(J - 1\right)^2,$$
(24)

with is clearly convex with respect to \mathcal{V} , provided that the material parameters $\{\mu_1, \mu_2, \lambda\}$ are positive. Notice that in (24), $|| \bullet ||$ represents the Frobenius norm of the argument \bullet . These can be related with the first and second

Lamé parameters in the origin (i.e. F = I, with $I \in \mathbb{R}^{3\times 3}$ the second order identity matrix), denoted as $\{\lambda_0, \mu_0\}$, respectively, according to the following expressions

$$\lambda_0 = \lambda + 2\mu_2; \qquad \mu_0 = \mu_1 + \mu_2. \tag{25}$$

For the electro-mechanical contribution $\Psi_{em}^{mac}(F, E_0)$, we consider the free energy function of an ideal dielectric elastomer, which is defined as

$$\Psi_{\rm em}^{\rm mac}(\boldsymbol{F}, \boldsymbol{E}_0) = -\frac{\varepsilon_r \varepsilon_0}{2J} ||\boldsymbol{H}\boldsymbol{E}_0||^2, \tag{26}$$

where ε_0 represents the electric permittivity of vacuum, being $\varepsilon_0 = 8.854 \times 10^{-12} \text{ Fm}^{-1}$, and ε_r the relative electric permittivity.

Remark 2. Notice that $\Psi_{em}^{mac}(F, E_0)$ is the dual of the following internal energy density function,

$$e_{\rm em}^{\rm mac}(\boldsymbol{F}, \boldsymbol{D}_0) = \frac{1}{2J\varepsilon_r\varepsilon_0} ||\boldsymbol{F}\boldsymbol{D}_0||^2, \tag{27}$$

where the relationship between both functions is given by the following Legendre transformation

$$\Psi_{\mathrm{em}}^{\mathrm{mac}}(\boldsymbol{F}, \boldsymbol{E}_0) = \inf_{\boldsymbol{D}_0} \{ e_{\mathrm{em}}^{\mathrm{mac}}(\boldsymbol{F}, \boldsymbol{D}_0) - \boldsymbol{E}_0 \cdot \boldsymbol{D}_0 \},$$

where (27) can be equivalently written as

$$e_{\mathrm{em}}^{\mathrm{mac}}(\boldsymbol{F}, \boldsymbol{D}_0) = \mathbb{W}(J, \boldsymbol{d}) = \frac{1}{2J\varepsilon_r\varepsilon_0}||\boldsymbol{d}||^2; \qquad \boldsymbol{d} = \boldsymbol{F}\boldsymbol{D}_0.$$

The authors (see Refs. [26–28]) have shown that the definition of $\mathbb{W}_{em}^{mac}(J, d)$ is indeed convex with respect to both of its arguments, provided that J > 0 (which is a physical condition preventing inter-penetrability of matter), namely the Hessian

$$\begin{bmatrix} \partial_{JJ}^2 \mathbb{W}_{em}^{\max}(J, d) & \partial_{Jd}^2 \mathbb{W}_{em}^{\max}(J, d) \\ \partial_{dJ}^2 \mathbb{W}_{em}^{\max}(J, d) & \partial_{dd}^2 \mathbb{W}_{em}^{\max}(J, d) \end{bmatrix}$$

if positive definite. In Refs. [26–28], the authors postulated the extension of polyconvexity to the field of electromechanics by permitting the internal energy density $e(F, D_0)$ to be convex with respect to any of the elements within the set $\{F, H, J, D_0, d\}$. The use of the free energy $\Psi_{em}^{mac}(F, E_0)$ in (26) may give the impression that it can jeopardise the polyconvexity condition, through the use of the minus sign. However, re-expressing $\Psi_{em}^{mac}(F, E_0)$ in terms of its dual $e_{em}^{mac}(F, D_0)$ dissipates any doubt. Therefore, $\Psi_{em}^{mac}(F, E_0)$ is indeed polyconvex in the context of electromechanics in the sense of [26–28].

2.5.2. Micromorphic free energy density

Following the work in [44], the expression adopted in this work for the micromorphic contribution $\Psi^{\text{mic}}(\mathcal{G})$ is the following

$$\Psi^{\text{mic}}(\boldsymbol{\mathcal{G}}) = \frac{\mu_0 l^2}{2} ||\boldsymbol{\mathcal{G}}||^2; \qquad ||\boldsymbol{\mathcal{G}}||^2 = \mathcal{G}_{iIJ} \mathcal{G}_{iIJ},$$
(28)

where $l \ge 0$ is the length-scale parameter and with μ_0 in Eq. (25). Notice that $||\mathcal{G}||^2$ in (28) represents the Frobenius norm of the third order tensor \mathcal{G} , and its expression has been shown in (28), where the use of repeated indices implies summation.

2.5.3. Scale-bridging free energy density

For the scale-bridging contribution $\Psi^{\text{scale}}(F, G)$, we use the same expression as that followed in Ref. [44], namely

$$\Psi^{\text{scale}}(\boldsymbol{F},\boldsymbol{G}) = \frac{\kappa}{2} ||\boldsymbol{F} - \boldsymbol{G}||^2, \tag{29}$$

where the material parameter κ in (29) is a penalty-like parameter coupling both macro- and micro-deformation gradients, i.e. F and G, respectively.

2.5.4. Flexoelectric free energy density

Finally, the flexoelectric contribution $\Psi^{\text{flexo}}(E_0, G, \mathcal{G})$, according to Ref. [44] adopts the following expression

$$\Psi^{\text{flexo}}(\boldsymbol{E}_0, \boldsymbol{G}, \boldsymbol{\mathcal{G}}) = \nu_f \left(\boldsymbol{G}^{-T} \boldsymbol{E}_0 \right) \cdot \left(\boldsymbol{\mathcal{G}} : \boldsymbol{I} \right); \qquad (\boldsymbol{\mathcal{G}} : \boldsymbol{I})_I = \mathcal{G}_{IJK} \delta_{JK} = \mathcal{G}_{IJJ}, \tag{30}$$

where the flexoelectric coefficient v_f is defined as

$$\nu_f = \varepsilon_0 l \bar{\nu}_f,\tag{31}$$

where $\bar{\nu}_f$ represents the dimensionless flexoelectric coefficient.

2.5.5. Specific form of the work conjugates and second order derivatives of the free energy density

The specification, through Sections 2.5.1–2.5.4, of each of the energetic contributions that the free energy density $\Psi(\mathcal{F})$ is decomposed into, permits to give the explicit definition of the work conjugates $\{P, D_0, \Sigma_G, \Sigma_G\}$ according to (14), namely

$$P = \partial_{F} \Psi(\mathcal{F}) = \mu_{1} F + \mu_{2} H \times F + f'(J) H + \frac{\varepsilon_{r} \varepsilon_{0}}{J} \left(\frac{1}{2J} || H E_{0} ||^{2} H - (H E_{0} \otimes E_{0}) \times F \right) + \kappa (F - G);$$

$$D_{0} = -\partial_{E_{0}} \Psi(\mathcal{F}) = \frac{\varepsilon_{r} \varepsilon_{0}}{J} H^{T} H E_{0} - \nu_{f} G^{-1} (\mathcal{G} : I);$$

$$\Sigma_{G} = \partial_{G} \Psi(\mathcal{F}) = \kappa (G - F) - \nu_{f} \left(G^{-T} E_{0} \otimes G^{-1} (\mathcal{G} : I) \right);$$

$$\Sigma_{\mathcal{G}} = \partial_{\mathcal{G}} \Psi(\mathcal{F}) = \mu_{0} l^{2} \mathcal{G},$$

(32)

where the cross product operation between second order tensors is defined as

$$(\boldsymbol{A} \times \boldsymbol{B})_{iI} = \mathcal{E}_{ijk} \mathcal{E}_{IJK} A_{jJ} B_{kK}; \qquad \forall \boldsymbol{A}, \boldsymbol{B} \in \mathbb{R}^{3 \times 3},$$
(33)

with $A \otimes B$ denoting the dyadic product for any *n*th order tensor A and B ($n \in \mathbb{N}$) and with \mathcal{E} the third order alternating tensor. In addition, we show in this section the various second partial derivatives of the free energy, namely $\partial_{XY}^2 \Psi(\mathcal{F})$ (for any $X, Y \in \mathcal{F}$), featuring in the different contributions of the linearised expressions in (17)–(19). In particular, for the specific choice of the constitutive model described in Section 2.5, these are as follows

$$\partial_{FF}^{2} \Psi(\mathcal{F}) = \mu_{1} \mathbb{I} + \mu_{2} F \times \mathbb{I} \times F + f''(J) H \otimes H + \mathbb{I} \times \left(\mu_{2} H + f'(J) F\right) + \varepsilon_{r} \varepsilon_{0} \left(-\frac{1}{J^{3}} ||HE_{0}||^{2} H \otimes H - \frac{1}{J} F \times \left(I \boxtimes (E_{0} \otimes E_{0}) \right) \times F + \frac{1}{J^{2}} \left(F \times (HE_{0} \otimes E_{0}) \right) \otimes H + \frac{1}{J^{2}} H \otimes \left(F \times (HE_{0} \otimes E_{0}) \right) + \mathbb{I} \times \left(-\frac{1}{J} HE_{0} \otimes E_{0} + \frac{1}{2J^{2}} ||HE_{0}||^{2} F \right) \right) + \kappa \mathbb{I};$$

$$\partial_{FE_{0}}^{2} \Psi(\mathcal{F}) = \varepsilon_{r} \varepsilon_{0} \left(-\frac{1}{J} F \times \left(HE_{0} \otimes I + H \boxtimes E_{0} \right) + \frac{1}{J^{2}} H \otimes H^{T} HE_{0} \right);$$

$$\partial^{2} \Psi_{FG}(\mathcal{F}) = -\kappa \mathbb{I},$$

$$(34)$$

with

$$(A \times A)_{iIjJ} = \mathcal{E}_{ipq} \mathcal{E}_{IPQ} A_{pP} \mathcal{A}_{qQjJ}; \qquad (A \times A)_{iIjJ} = \mathcal{E}_{jpq} \mathcal{E}_{JPQ} \mathcal{A}_{iIpP} A_{qQ} (\mathbb{I})_{iIjJ} = \delta_{ij} \delta_{IJ}; \qquad (A \boxtimes B)_{iIjJ} = A_{ij} B_{IJ},$$
(35)

 $\forall A, B \in \mathbb{R}^{3 \times 3}$ and $\mathcal{A} \in \mathbb{R}^{3 \times 3 \times 3 \times 3}$. In addition, the derivatives $\partial_{E_0 E_0}^2 \Psi(\mathcal{F})$, $\partial_{E_0 G}^2 \Psi(\mathcal{F})$ and $\partial_{E_0 \mathcal{G}}^2 \Psi(\mathcal{F})$ are as follows

$$\partial_{E_0 E_0}^2 \Psi(\mathcal{F}) = -\frac{\varepsilon_r \varepsilon_0}{2J} H^T H E_0; \quad \partial_{E_0 G}^2 \Psi(\mathcal{F}) = -\nu_f G^{-1} \otimes \left(G^{-1} \left(\mathcal{G} : I \right) \right); \quad \partial_{E_0 \mathcal{G}}^2 \Psi(\mathcal{F}) = -\nu_f G^{-1} \otimes I.$$
(36)



Fig. 2. The FM-Elasticity continuum formulation considered and its particularisation to other continuum approaches. *Source:* Figure adapted from [44].

Furthermore, the derivatives $\partial_{GG}^2 \Psi(\mathcal{F})$ and $\partial_{GG}^2 \Psi(\mathcal{F})$ can be obtained as

$$\partial_{GG}^{2}\Psi(\mathcal{F}) = \kappa \mathbb{I} + \nu_{f} \left(G^{-T} \odot \left(\left(G^{-1}(\mathcal{G}:I) \right) \otimes \left(G^{-1}E_{0} \right) \right) + \left(G^{-1}E_{0} \right) \otimes G^{-1} \otimes \left(G^{-1}(\mathcal{G}:I) \right) \right);$$

$$\partial_{GG}^{2}\Psi(\mathcal{F}) = -\nu_{f} \left(G^{-T}E_{0} \right) \otimes G^{-1} \otimes I,$$
(37)

with $(\mathbf{A} \odot \mathbf{B})_{iIjJ} = \delta_{iJ}\delta_{Ij}, \forall \mathbf{A}, \mathbf{B} \in \mathbb{R}^{3\times 3}$. Finally, the last derivative featuring in (A.6) is

$$\partial_{\mathcal{GG}}^2 \Psi(\mathcal{F}) = \mu_0 l^2 \widetilde{\mathbb{I}}; \qquad \left(\widetilde{\mathbb{I}} \right)_{iPQjRS} = \delta_{ij} \delta_{PR} \delta_{QS}. \tag{38}$$

2.6. The flexoelectric micromorphic approach in the context of microcontinuum theories

Fig. 2 showcases the convenience and flexibility of the micromorphic approach adopted in this manuscript. As discussed in Ref. [44], one of the great advantages of the micromorphic approach considered, denoted as *FM*-*Elasticity* (which stands for coupled Flexoelectricity and Micromorphic Elasticity), resides in the fact that it can be particularised, by tuning some of the material parameters featuring in the different energy contributions described in the preceding sections, into a wide range of less generic modelling approaches. For instance, for the limiting case when the scale bridging coefficient κ in (29) is $\kappa \to \infty$, the formulation reduces to Flexoelectric Gradient Elasticity (*FG-Elasticity*) [41], where the micro- and macroscopic deformation gradient tensors *G* and *F* coincide, namely F = G, and hence, $\mathcal{G} = \nabla_0 F$. *FG-Elasticity* permits the use of mixed-type variational formulations, but it can also be modelled through a two field variational principle (with only { ϕ , φ } as unknown fields), forcing a higher regularity for the discretisation of the field ϕ , requiring $C^1(\mathcal{B}_0)$ -continuous finite element basis functions. The latter is precisely the formulation pursued in [40,41].

Furthermore, setting the flexoelectric coefficient $\bar{v}_f = 0$ in (30)–(31), *FM-Elasticity* reduces to a family of continuum formulations devoid from flexoelectricity. The most generic of them is *EM-Elasticity* (i.e. Electro Micromorphic Elasticity). The latter can reduce into other less generic continuum approaches such as Electromechanics Gradient Elasticity (*EG-Elasticity*) and Electromechanics Elasticity (*E-Elasticity*), and into formulations excluding at all any source of electromechanical coupling, such as Micromorphic Elasticity (*M-Elasticity*), Gradient Elasticity (*G-Elasticity*), and finally, *Elasticity*.

3. Optimisation via the SIMP approach

3.1. Introduction

As customary in density-based topology optimisation (see [45]), we seek the optimised design, in this specific case, of a flexoelectric energy harvester, through the introduction of the *design density field* $\rho(X) \in [0, 1]$, which

represents a continuous scalar field introduced in order to differentiate solid from void regions in \mathcal{B}_0 . In order to prevent spurious results, we make use of standard filtering techniques [85,86] such as the conic filter, which can be stated as the convolution product

$$\tilde{\rho}(\boldsymbol{X}) = (\rho * \Delta)(\boldsymbol{X}) = \int_{\mathcal{B}_0} \rho(\boldsymbol{X}') \Delta(||\boldsymbol{X} - \boldsymbol{X}'||) \, d\boldsymbol{V},\tag{39}$$

where $\Delta : \mathbb{R}_+ \to \mathbb{R}_+$ is the so-called convolution kernel, $\tilde{\rho}(X)$ is referred to as the *filtered density field*, and with $\Delta(r) = \max\{0, 1-r/R\}$, being *R* the filter radius. Alternatively, it is possible to use the Helmholtz' filter, resulting from the solution of the following PDE

$$\tilde{\rho} - R^2 \Delta_0 \tilde{\rho} = \rho; \qquad \text{on } \mathcal{B}_0;
\nabla_0 \tilde{\rho} \cdot N = 0; \qquad \text{on } \partial \mathcal{B}_0,$$
(40)

where *R* represents the filter radius and where $\Delta_0(\bullet)$ represents the material or Lagrangian Laplacian operator. In addition, we use the classical smoothed Heaviside projection function proposed in [87]

$$\hat{\rho}(X) = \frac{\tanh(\beta\eta) + \tanh(\beta(\tilde{\rho}(X) - \eta))}{\tanh(\beta\eta) + \tanh(\beta(1 - \eta))},\tag{41}$$

where $\hat{\rho}(X)$ is known as the *physical density field* and β and η are parameters carefully selected and updated throughout the optimisation [87]. The field $\hat{\rho} : \mathcal{B}_0 \to [0, 1]$ is used in order to interpolate the free energy density of the body $\Psi(X, \mathcal{F})$, which depends upon each macroscopic point $X \in \mathcal{B}_0$ through $\hat{\rho}(X)$, namely

$$\Psi(X, \mathcal{F}) = \Psi(\hat{\rho}(X), \mathcal{F}); \qquad \mathcal{F} = \{F, E_0, G, \mathcal{G}\}.$$
(42)

The additive decomposition of the free energy considered, shown in Eqs. (21) and (22), permits to establish an independent energy interpolation scheme for each of the five energetic contributions considered, namely

$$\Psi_{m}^{\text{mac}}(X, F) = \Psi_{m}^{\text{mac}}(\hat{\rho}(X), F); \qquad \Psi_{em}^{\text{mac}}(X, F, E_{0}) = \Psi_{em}^{\text{mac}}(\hat{\rho}(X), F, E_{0}); \\
\Psi_{m}^{\text{mic}}(X, \mathcal{G}) = \bar{\Psi}^{\text{mic}}(\hat{\rho}(X), \mathcal{G}); \qquad \Psi^{\text{scale}}(X, F, G) = \bar{\Psi}^{\text{scale}}(\hat{\rho}(X), F, G); \quad (43)$$

$$\Psi^{\text{flexo}}(X, E_{0}, G, \mathcal{G}) = \bar{\Psi}^{\text{scale}}(\hat{\rho}(X), E_{0}, G, \mathcal{G}).$$

The functional dependence of each of the five contributions in Eq. (43) with respect to the field $\hat{\rho}(X)$, is dictated by the choice of the energy interpolation scheme considered. Irrespectively of that choice, the five energy interpolation schemes in (43) must verify that, for the limiting case $\hat{\rho} = 1$, the interpolated energy must coincide with that of the solid material considered. On the other hand, for the limiting case $\hat{\rho} = 0$, the interpolated energetic contributions are forced to coincide with a small fraction (controlled by a parameter $\alpha \ll 1$) of that of the solid material (a zero contribution $\bar{\Psi}_{em}^{mac}$, however, we impose that for the limiting case $\hat{\rho} = 0$, the energetic contribution must coincide with that of vacuum, denoted as Ψ_{em}^{vac} , and defined as

$$\Psi_{\rm em}^{\rm vac} = -\frac{\varepsilon_0}{2J} ||\boldsymbol{H}\boldsymbol{E}_0||^2.$$
(44)

All of that can be mathematically stated below as

$$\bar{\Psi}_{m}^{mac}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F}) = \begin{cases} \Psi_{m}^{mac}(\boldsymbol{F}); & \hat{\rho} = 1; \\ \alpha \Psi_{m}^{mac}(\boldsymbol{F}); & \hat{\rho} = 0 \end{cases}; \\
\bar{\Psi}_{em}^{mac}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F}, \boldsymbol{E}_{0}) = \begin{cases} \Psi_{em}^{mac}(\boldsymbol{F}, \boldsymbol{E}_{0}); & \hat{\rho} = 1; \\ \Psi_{em}^{vac}(\boldsymbol{F}, \boldsymbol{E}_{0}); & \hat{\rho} = 0 \end{cases}; \\
\bar{\Psi}_{m}^{mic}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{\mathcal{G}}) = \begin{cases} \Psi_{mic}^{mac}(\boldsymbol{\mathcal{G}}); & \hat{\rho} = 1; \\ \alpha \Psi^{mic}(\boldsymbol{\mathcal{G}}); & \hat{\rho} = 0 \end{cases}; \\
\bar{\Psi}_{em}^{scale}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F}, \boldsymbol{G}) = \begin{cases} \Psi_{em}^{scale}(\boldsymbol{F}, \boldsymbol{G}); & \hat{\rho} = 1; \\ \alpha \Psi^{scale}(\boldsymbol{F}, \boldsymbol{G}); & \hat{\rho} = 0 \end{cases}; \\
\bar{\Psi}_{em}^{scale}(\boldsymbol{F}, \boldsymbol{G}); & \hat{\rho} = 1; \\
\alpha \Psi^{scale}(\boldsymbol{F}, \boldsymbol{G}); & \hat{\rho} = 1; \\
\alpha \Psi^{scale}(\boldsymbol{F}, \boldsymbol{G}); & \hat{\rho} = 0 \end{cases}; \\
\bar{\Psi}_{em}^{flexo}(\boldsymbol{E}_{0}, \boldsymbol{G}, \boldsymbol{\mathcal{G}}) = \begin{cases} \Psi_{em}^{flexo}(\boldsymbol{E}_{0}, \boldsymbol{G}, \boldsymbol{\mathcal{G}}); & \hat{\rho} = 1; \\
\alpha \Psi^{flexo}(\boldsymbol{E}_{0}, \boldsymbol{G}, \boldsymbol{\mathcal{G}}); & \hat{\rho} = 0 \end{cases}; \end{cases}$$

Unfortunately, the limiting values of $\bar{\Psi}_{m}^{mac}(\hat{\rho}(X), F)$ and $\bar{\Psi}_{em}^{mac}(\hat{\rho}(X), F, E_{0})$ in (45) for $\hat{\rho} = 0$ can induce artificial numerical instabilities. In order to remedy this shortcoming, it is customary [72,73] to replace them with the following more stable definitions

$$\bar{\Psi}_{m}^{\text{mac}}(\hat{\rho}(X), F)\big|_{\hat{\rho}=0} = \Psi_{m}^{\text{void}}(F) = \frac{\alpha}{2} (F - I) : \partial_{FF}^{2} \Psi_{m}^{\text{mac}}(F)\big|_{F=I} : (F - I);$$

$$\bar{\Psi}_{em}^{\text{vac}}(\hat{\rho}(X), F, E_{0})\big|_{\hat{\rho}=0} = \Psi_{em}^{\text{void}}(F, E_{0}) = -\frac{\varepsilon_{0}}{2} ||E_{0}||^{2},$$
(46)

where it can be seen that the electromechanical contribution in the void region $\Psi_{em}^{void}(F, E_0)$ is indeed a particularisation of that of the vacuum $\Psi_{em}^{vac}(F, E_0)$ for the case H = I. Inspired by the SIMP method [45], a possible energy interpolation scheme for each of the five energy contributions that complies with the extreme or limiting cases established in (45) and (46) is the following

$$\bar{\Psi}_{m}^{mac}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F}) = (\hat{\rho}(\boldsymbol{X}))^{p_{m}} \Psi^{mac}(\boldsymbol{F}) + [1 - (\hat{\rho}(\boldsymbol{X}))^{p_{m}}] \Psi_{m}^{void}(\boldsymbol{F});$$

$$\bar{\Psi}_{em}^{mac}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F}, \boldsymbol{E}_{0}) = (\hat{\rho}(\boldsymbol{X}))^{p_{e}} \Psi^{mac}(\boldsymbol{F}, \boldsymbol{E}_{0}) + [1 - (\hat{\rho}(\boldsymbol{X}))^{p_{e}}] \Psi_{em}^{void}(\boldsymbol{E}_{0});$$

$$\bar{\Psi}^{mic}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{\mathcal{G}}) = (\alpha + (1 - \alpha) (\hat{\rho}(\boldsymbol{X}))^{p_{mic}}) \Psi^{mic}(\boldsymbol{\mathcal{G}});$$

$$\bar{\Psi}^{scale}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F}, \boldsymbol{G}) = (\alpha + (1 - \alpha) (\hat{\rho}(\boldsymbol{X}))^{p_{scale}}) \Psi^{scale}(\boldsymbol{F}, \boldsymbol{G});$$

$$\bar{\Psi}^{flexo}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{E}_{0}, \boldsymbol{G}, \boldsymbol{\mathcal{G}}) = (\alpha + (1 - \alpha) (\hat{\rho}(\boldsymbol{X}))^{p_{flexo}}) \Psi^{flexo}(\boldsymbol{E}_{0}, \boldsymbol{G}, \boldsymbol{\mathcal{G}}),$$
(47)

where the five parameters { p_m , p_e , p_{mic} , p_{scale} , p_{flexo} } featuring in the five interpolated energy contributions in (47), namely { $\bar{\Psi}_m^{mac}(\hat{\rho}(X), F)$, $\bar{\Psi}_{em}^{mac}(\hat{\rho}(X), F, E_0)$, $\bar{\Psi}^{mic}(\hat{\rho}(X), G)$, $\bar{\Psi}^{scale}(\hat{\rho}(X), F, G)$, $\bar{\Psi}^{flexo}(\hat{\rho}(X), E_0, G, G)$ }, play the role of the penalising exponent in the SIMP method in the context of elasticity, where only one exponent is needed.

3.2. The choice of the objective function

In this paper, we seek to obtain the optimised design of a flexoelectro energy harvester \mathcal{B}_0 . Specifically, we aim at optimising an objective function embedding the energy harvesting properties of the flexoelectric material. With that objective in mind, previous references in the field of topology optimisation of flexoelectric materials [77,78], identify the objective function as the efficiency, denoted as $\mathcal{J}_{\eta}(\hat{\rho}, \phi, \varphi)$, and defined as the (negative) ratio between the mechanical and electromechanical contributions of the macroscopic energy contribution, namely

$$\mathcal{J}_{\eta}(\hat{\rho}, \boldsymbol{\phi}, \varphi) = \int_{\mathcal{B}_0} -\frac{\bar{\Psi}_{\mathrm{m}}^{\mathrm{mac}}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F})}{\Psi_{\mathrm{em}}^{\mathrm{mac}}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F}, \boldsymbol{E}_0)} \, dV.$$
(48)

However, as it will be shown in the numerical examples section, identification of the efficiency in (48) as the objective function of the problem leads to designs characterised by a poor structural performance, with hinges where high stress concentration occurs. This mechanical deficiency is indeed promoted by the inherent flexoelectric nature of the material, which promotes the development of hinges with highly localised large strain gradients. The latter leads to an extremely rapid growth of the electric energy generated when compared with the associated mechanical energy decay, yielding extremely efficient designs according to the metric chosen, but inappropriate from a structural/mechanical standpoint. In order to remedy the above shortcoming, it is possible to propose an objective function defined as a combination of the efficiency in (48) and a p-norm aggregation/regularisation function of the Von Mises stress distribution within the material.¹ The latter, denoted herein as $\mathcal{J}_{\tau}(\hat{\rho}, \phi)$, can be defined as

$$\mathcal{J}_{\tau}(\hat{\rho}, \boldsymbol{\phi}) = \left(\sum_{e=1}^{N} \left(\frac{\int_{\mathcal{B}_{0}^{e}} \tau(\hat{\rho}(\boldsymbol{X}), \boldsymbol{\phi}) \, dV}{\tau_{\max} \int_{\mathcal{B}_{0}^{e}} \, dV}\right)^{\gamma}\right)^{1/\gamma} \approx \max_{e} \left\{\frac{\int_{\mathcal{B}_{0}^{e}} \tau(\hat{\rho}(\boldsymbol{X}), \boldsymbol{\phi}) \, dV}{\tau_{\max} \int_{\mathcal{B}_{0}^{e}} \, dV}\right\},\tag{49}$$

where the quality of the approximation of the maximum operator is clearly affected by the choice of the exponent $\gamma \in \mathbb{R}^+$. Typically, whilst high values of γ yield a better approximation, they might compromise the stability of the

¹ Notice that alternative aggregation techniques such as p-mean or Kreisselmeier–Steinhauser can be used for the global measure $\mathcal{J}_{\tau}(\hat{\rho}, \boldsymbol{\phi}, \varphi)$.

optimisation algorithm due to the high nonlinearity of the regularised functional $\mathcal{J}_{\tau}(\hat{\rho}, \phi)$. Furthermore, the local stress measure $\tau(\hat{\rho}(X), \phi)$ in (49) is defined as

$$\tau(\hat{\rho}(X), \boldsymbol{\phi}) = \sqrt{3} ||\boldsymbol{\tau}^{\text{dev}}(\hat{\rho}(X), \boldsymbol{\phi})||^2,$$
(50)

where $\tau^{\text{dev}}(\hat{\rho}(X), \phi)$ is the deviatoric part of the interpolated Kirchhoff stress tensor $\tau(\hat{\rho}(X), \phi)$, namely

$$\boldsymbol{\tau}^{\text{dev}}(\hat{\rho}(X),\boldsymbol{\phi}) = \mathbb{P}^{\text{dev}}: \boldsymbol{\tau}(\hat{\rho}(X),\boldsymbol{\phi}); \qquad \boldsymbol{\tau} = \partial_F \,\hat{\Psi}_{\text{m}}^{\text{mac}}(\hat{\rho}(X),F)F^T, \tag{51}$$

where the fourth order tensor \mathbb{P}^{dev} is defined as

$$\mathbb{P}^{\text{dev}} = \mathbb{I}^{\text{sym}} - \frac{1}{3} \mathbf{I} \otimes \mathbf{I}; \qquad \left(\mathbb{I}^{\text{sym}}\right)_{ijkl} = \frac{1}{2} \left(\delta_{ik} \delta_{jl} + \delta_{il} \delta_{jk}\right), \tag{52}$$

where δ_{ij} refers to the -ij component of the Kronecker delta tensor. Notice in above Eq. (51) that the interpolated energy $\tilde{\Psi}_{m}^{mac}(\hat{\rho}(X), F)$ is defined similarly to the mechanical contribution of the macroscopic interpolated energy $\bar{\Psi}_{m}^{mac}(\hat{\rho}(X), F)$ in (47), i.e.

$$\hat{\Psi}_{\mathrm{m}}^{\mathrm{mac}}(\hat{\rho}(X), F) = \left(\hat{\rho}(X)\right)^{p_{\tau}} \Psi_{\mathrm{m}}^{\mathrm{mac}}(F) + \left[1 - \left(\hat{\rho}(X)\right)^{p_{\tau}}\right] \Psi_{\mathrm{m}}^{\mathrm{void}}(F)$$
(53)

In fact, the only difference between $\hat{\Psi}_{m}^{mac}(\hat{\rho}(X), F)$ in (53) and $\bar{\Psi}_{m}^{mac}(\hat{\rho}(X), F)$ in (47) resides in the choice of the penalising exponent, which is either p_{τ} or p_{m} , respectively. This difference is motivated by the possible development of classical shortcomings, i.e. stress singularity, associated with stress minimisation or stress constrained topology optimisation [79]. In order to remedy these, it is customary to consider a value of the penalising exponent of $p_{\tau} = 0.5$ [79]. Having introduced both functionals $\mathcal{J}_{\eta}(\hat{\rho}, \boldsymbol{\phi}, \varphi)$ in (48) and $\mathcal{J}_{\tau}(\hat{\rho}, \boldsymbol{\phi})$ in (49), it is now possible to define the objective function $\mathcal{J}(\hat{\rho}, \boldsymbol{\phi}, \varphi)$ as a weighted combination of the two according to

$$\mathcal{J}(\hat{\rho}, \boldsymbol{\phi}, \varphi) = \frac{\mathcal{J}_{\eta}(\hat{\rho}, \boldsymbol{\phi}, \varphi)}{\mathcal{J}_{\eta_0}(\hat{\rho}, \boldsymbol{\phi}, \varphi)} + \omega \frac{\mathcal{J}_{\tau}(\hat{\rho}, \boldsymbol{\phi})}{\mathcal{J}_{\tau_0}(\hat{\rho}, \boldsymbol{\phi})},\tag{54}$$

where \mathcal{J}_{η_0} and \mathcal{J}_{τ_0} refer to initial values of \mathcal{J}_{η} and \mathcal{J}_{τ} , respectively, i.e. their values at the initial topology optimisation iteration. In Eq. (54), ω represents the weight associated with the stress-based objective functional $\mathcal{J}_{\tau}(\hat{\rho}, \phi)$. Alternative definitions for the objective function $\mathcal{J}(\hat{\rho}, \phi, \varphi)$ can be considered, as the one presented below

$$\mathcal{J}(\hat{\rho}, \boldsymbol{\phi}, \varphi) = \log_{10} \mathcal{J}_{\eta}(\hat{\rho}, \boldsymbol{\phi}, \varphi) + \omega \frac{\mathcal{J}_{\tau}(\hat{\rho}, \boldsymbol{\phi})}{\mathcal{J}_{\tau_0}(\hat{\rho}, \boldsymbol{\phi})}.$$
(55)

Remark 3. The Von Mises-based functional \mathcal{J}_{τ} in (49) has been introduced with the aim of reducing high stress values which, in turn, prevents the development of hinges and hence, of extremely localised deformations. This, therefore, can help limiting the value of the electric field attained through the inherent flexoelectric behaviour of the material. As a result, a reduction of high electric field concentrations is achieved in an indirect manner. Alternative functionals to \mathcal{J}_{τ} could have been defined incorporating as a constraint, or in a multi-objective fashion, the maximum value of the norm of the electric field attained through flexoelectricity. Through this approach, a direct control on the maximum electric field can also be obtained. Either of above two approaches would permit to incorporate important limiting physical phenomena inherent to dielectric elastomers, such as electrical breakdown [88], into the optimisation problem.

3.3. The optimisation Lagrangian L: descend direction

The optimisation problem can be finally formulated as

$$\begin{cases} \min_{\rho(X)} \mathcal{J}(\rho, \phi, \varphi), \\ \text{State equations (1), (4), (5)-(8);} \\ \text{Constitutive model (21)-(30);} \\ \text{Energy interpolation scheme in (47);} \\ 0 \le \rho(X) \le 1; \quad \rho(X) \in L^2(\mathcal{B}_0); \\ \text{Volume constraint } g_{\mathcal{B}_0}(\rho) = \frac{\int_{\mathcal{B}_0} \hat{\rho}(X) dV}{\int_{\mathcal{B}_0} dV} \le f. \end{cases}$$

(56)

Associated with the minimisation problem (56), the following Lagrangian functional \mathcal{L} (excluding the volume constraint and the upper and lower bound constraints for the sake of simplicity) is defined

$$\mathcal{L}(\rho, \mathcal{U}, p_{\mathcal{U}}) = \mathcal{J}(\rho, \phi, \varphi) - \underbrace{\left(D\Pi(\rho, \mathcal{U})[p_{\phi}] + D\Pi(\rho, \mathcal{U})[p_{\varphi}] + D\Pi(\rho, \mathcal{U})[p_{G}]\right)}_{=0.PDE-constraints},$$
(57)

with $\mathcal{U} = \{\phi, \varphi, G\}$ and with the adjoint states encapsulated in $p_{\mathcal{U}} = \{p_{\phi}, p_{\varphi}, p_{G}\} \in \mathbb{V}_{0}^{\phi} \times \mathbb{V}_{0}^{\varphi} \times \mathbb{V}_{0}^{G}$. In (57), the difference between $D\Pi(\rho, \mathcal{U})[p_{\phi}]$, $D\Pi(\rho, \mathcal{U})[p_{\varphi}]$ and $D\Pi(\rho, \mathcal{U})[p_{G}]$ and $D\Pi(\mathcal{U})[p_{\phi}]$, $D\Pi(\mathcal{U})[p_{\phi}]$, and $D\Pi(\mathcal{U})[p_{G}]$ in Eq. (13) resides in the replacement of the free energy $\Psi(\mathcal{F})$ with its interpolated counterpart $\bar{\Psi}(\hat{\rho}(X), \mathcal{F})$ in (42), (43) and (47), namely

$$D\Pi(\rho, \mathcal{U})[\boldsymbol{p}_{\phi}] = \int_{\mathcal{B}_{0}} \partial_{F} \bar{\Psi}\left(\hat{\rho}(\boldsymbol{X}), \mathcal{F}\right) : \nabla_{0}\boldsymbol{p}_{\phi} \, dV - \int_{\mathcal{B}_{0}} \boldsymbol{f}_{0} \cdot \boldsymbol{p}_{\phi} - \int_{\partial_{t_{0}}\mathcal{B}_{0}} \boldsymbol{t}_{0} \cdot \boldsymbol{p}_{\phi} \, dA;$$

$$D\Pi(\rho, \mathcal{U})[\boldsymbol{p}_{\varphi}] = -\int_{\mathcal{B}_{0}} \partial_{E_{0}} \bar{\Psi}\left(\hat{\rho}(\boldsymbol{X}), \mathcal{F}\right) \cdot \nabla_{0}\boldsymbol{p}_{\varphi} \, dV + \int_{\mathcal{B}_{0}} \rho_{0}\boldsymbol{p}_{\varphi} + \int_{\partial_{\omega_{0}}\mathcal{B}_{0}} \omega_{0}\boldsymbol{p}_{\varphi} \, dA;$$

$$D\Pi(\rho, \mathcal{U})[\boldsymbol{p}_{G}] = \int_{\mathcal{B}_{0}} \partial_{G} \bar{\Psi}\left(\hat{\rho}(\boldsymbol{X}), \mathcal{F}\right) : \boldsymbol{p}_{G} \, dV + \int_{\mathcal{B}_{0}} \partial_{\mathcal{G}} \bar{\Psi} : \nabla_{0}\boldsymbol{p}_{G} \, dV - \int_{\partial_{T_{0}}\mathcal{B}_{0}} \boldsymbol{T}_{0} : \boldsymbol{p}_{G} \, dA.$$
(58)

In the context of gradient optimisation, the directional derivative of the $\mathcal{L}(\rho, \mathcal{U}, p_{\mathcal{U}})$ (alternatively called as *descend direction*), permits the update of the density field in subsequent topology optimisation iterations. The objective of this section is therefore, to obtain the expression of the descend direction of the Lagrangian $\mathcal{L}(\rho, \mathcal{U}, p_{\mathcal{U}})$ in (57). Notice however that, as discussed in Section 3.2, the objective function $\mathcal{J}(\hat{\rho}, \phi, \varphi)$, featuring in the Lagrangian $\mathcal{L}(\rho, \mathcal{U}, p_{\mathcal{U}})$ in (58), comprises a possibly nonlinear balance between the functionals $\mathcal{J}_{\eta}(\hat{\rho}, \phi, \varphi)$ and $\mathcal{J}_{\tau}(\hat{\rho}, \phi)$ in Eqs. (48) and (49), respectively. For instance, for the choice in Eq. (55), the descend direction will depend upon the descend directions of the each individual contribution $\mathcal{J}_{\eta}(\hat{\rho}, \phi, \varphi)$ and $\mathcal{J}_{\tau}(\hat{\rho}, \phi)$ according to

$$D\mathcal{L}(\rho, \phi, \varphi)[\Delta\rho] = \frac{\log_{10}e}{\log_{10}\mathcal{J}_{\eta}(\hat{\rho}, \phi, \varphi)} D\mathcal{L}_{\eta}[\Delta\rho] + \omega \frac{D\mathcal{L}_{\tau}(\rho, \phi)[\Delta\rho]}{\mathcal{J}_{\tau_0}(\hat{\rho}, \phi)},$$
(59)

where the individual descend directions $D\mathcal{L}_{\eta}[\Delta\rho]$ and $D\mathcal{L}_{\tau}(\rho, \phi)[\Delta\rho]$ in (59) can be obtained by defining their Lagrangian functionals $\mathcal{L}_{\eta}(\rho, \mathcal{U}, p_{\mathcal{U}})$ and $\mathcal{L}_{\tau}(\rho, \mathcal{U}, p_{\mathcal{U}})$ in a similar fashion as in (57), namely

$$\mathcal{L}_{X}(\rho, \mathcal{U}, \boldsymbol{p}_{\mathcal{U}}) = \mathcal{J}_{X}(\rho, \boldsymbol{\phi}, \varphi) - D\Pi(\rho, \mathcal{U})[\boldsymbol{p}_{\boldsymbol{\phi}}] - D\Pi(\rho, \mathcal{U})[\boldsymbol{p}_{\varphi}] - D\Pi(\rho, \mathcal{U})[\boldsymbol{p}_{\boldsymbol{G}}], \quad X = \{\eta, \tau\}$$
(60)

For each of the two Lagrangian functionals $\mathcal{L}_X(\rho, \mathcal{U}, p_{\mathcal{U}})$, their descend directions can be obtained as

$$D\mathcal{L}_{X}(\rho, \phi, \varphi)[\Delta\rho] = D\mathcal{J}_{X}(\hat{\rho}, \phi, \varphi)[D\hat{\rho}[\Delta\rho]] - \int_{\mathcal{B}_{0}} \left[\nabla_{0} \boldsymbol{p}_{\phi} : \partial_{\hat{\rho}} \left(\partial_{F} \bar{\Psi}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{\mathcal{F}}) \right) \right] D\hat{\rho}[\Delta\rho] dV + \int_{\mathcal{B}_{0}} \left[\nabla_{0} \boldsymbol{p}_{\varphi} \cdot \partial_{\hat{\rho}} \left(\partial_{E_{0}} \bar{\Psi}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{\mathcal{F}}) \right) \right] D\hat{\rho}[\Delta\rho] dV - \int_{\mathcal{B}_{0}} \left[\nabla_{0} \boldsymbol{p}_{G} : \partial_{\hat{\rho}} \left(\partial_{G} \bar{\Psi}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{\mathcal{F}}) \right) \right] D\hat{\rho}[\Delta\rho] dV - \int_{\mathcal{B}_{0}} \left[\nabla_{0} \boldsymbol{p}_{\mathcal{G}} : \partial_{\hat{\rho}} \left(\partial_{\mathcal{G}} \bar{\Psi}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{\mathcal{F}}) \right) \right] D\hat{\rho}[\Delta\rho] dV,$$

$$(61)$$

with $D\hat{\rho}[\Delta\rho] = \frac{\partial\hat{\rho}}{\partial\hat{\rho}}\frac{\partial\rho}{\partial\rho}\Delta\rho$, where $\frac{\partial\hat{\rho}}{\partial\hat{\rho}}$ and $\frac{\partial\hat{\rho}}{\partial\rho}$ are obtained from (39) and (41), respectively. The explicit expression of the directional derivatives $D\mathcal{J}_X(\hat{\rho}, \phi, \varphi)[D\hat{\rho}[\Delta\rho]]$ featuring in (61) for the cases where $X = \{\eta, \tau\}$ will be shown in Sections 3.3.1 and 3.3.2, respectively.

Clearly, in order to obtain the descend direction $D\mathcal{L}_X(\rho, \mathcal{U}, p_{\mathcal{U}})[\Delta\rho]$, it is necessary first to obtain the values of the unknown fields \mathcal{U} and those of the adjoint states $p_{\mathcal{U}}$. The first, namely the unknown fields \mathcal{U} , are obtained from

the optimality condition of the Lagrangian $\mathcal{L}_X(\rho, \mathcal{U}, p_{\mathcal{U}})$ with respect to $\delta p_{\mathcal{U}}$ (alternatively called as the *forward problem*), namely

$$D\mathcal{L}_{X}(\rho, \mathcal{U}, \boldsymbol{p}_{\mathcal{U}})[\delta \boldsymbol{p}_{\mathcal{U}}] = -D\Pi(\rho, \mathcal{U})[\delta \boldsymbol{p}_{\mathcal{U}}] = 0 \Rightarrow \begin{cases} D\Pi(\rho, \mathcal{U})[\delta \boldsymbol{p}_{\phi}] = 0; \\ D\Pi(\rho, \mathcal{U})[\delta \boldsymbol{p}_{\phi}] = 0; \end{cases} \Rightarrow \text{Get fields } \mathcal{U} = \{\boldsymbol{\phi}, \varphi, \boldsymbol{G}\} \\ D\Pi(\rho, \mathcal{U})[\delta \boldsymbol{p}_{G}] = 0, \end{cases}$$

$$(62)$$

where the expression for $D\Pi(\rho, \mathcal{U})[\delta p_{\mathcal{U}}]$ are the same as those for $D\Pi(\rho, \mathcal{U})[p_{\mathcal{U}}]$ in (61), exchanging $\delta p_{\mathcal{U}}$ with $p_{\mathcal{U}}$. Clearly, the optimality conditions for both Lagrangians are exactly identical, namely $D\mathcal{L}_{\eta}(\rho, \mathcal{U}, p_{\mathcal{U}})[\delta p_{\mathcal{U}}] = D\mathcal{L}_{\tau}(\rho, \mathcal{U}, p_{\mathcal{U}})[\delta p_{\mathcal{U}}]$, and therefore, the forward problem needs to be solved only once. The adjoint fields $p_{\mathcal{U}}$ can be obtained from the optimality condition of the Lagrangian $\mathcal{L}_{X}(\rho, \mathcal{U}, p_{\mathcal{U}})$ with respect to $\delta\mathcal{U}$ (alternatively called as the *adjoint problem*), namely

$$D\mathcal{L}_{X}(\rho, \mathcal{U}, \boldsymbol{p}_{\mathcal{U}})[\delta\mathcal{U}] = 0 \Rightarrow \begin{cases} D\mathcal{J}_{X}(\hat{\rho}, \boldsymbol{\phi}, \varphi)[\delta\boldsymbol{\phi}] - D^{2}\Pi(\rho, \mathcal{U})[\delta\boldsymbol{\phi}; \boldsymbol{p}_{\boldsymbol{\phi}}] = 0; \\ -D^{2}\Pi(\rho, \mathcal{U})[\delta\boldsymbol{G}; \boldsymbol{p}_{\boldsymbol{\phi}}] - D^{2}\Pi(\rho, \mathcal{U})[\delta\boldsymbol{G}; \boldsymbol{p}_{\boldsymbol{\phi}}] - D^{2}\Pi(\rho, \mathcal{U})[\delta\boldsymbol{G}; \boldsymbol{p}_{\boldsymbol{\phi}}] - D^{2}\Pi(\rho, \mathcal{U})[\delta\boldsymbol{G}; \boldsymbol{p}_{\boldsymbol{\phi}}] = 0, \end{cases}$$

Getfields $p_{\mathcal{U}} = \{p_{\phi}, p_{\varphi}, p_{G}\}$

(63)

where we have made use of the Schwartz theorem that permits to use $D^2 \Pi(\rho, \mathcal{U})[\delta \mathcal{U}; p_{\mathcal{U}}]$ in (63) instead of $D^2 \Pi(\rho, \mathcal{U})[p_{\mathcal{U}}; \delta \mathcal{U}]$, due to their equivalence. Notice in above equation that the expressions for $D^2 \Pi(\rho, \mathcal{U})[\delta \mathcal{U}; p_{\mathcal{U}}]$ in (63) are in fact the same as the linearised expressions in Eqs. (17)–(19). Finally, the explicit expressions for $D\mathcal{J}_X(\hat{\rho}, \phi, \varphi)[\delta \phi]$ for $X = \{\eta, \tau\}$ will be shown in Sections 3.3.1 and 3.3.2. The different expressions for both cases where $X = \{\eta, \tau\}$ entails that the adjoints states $p_{\mathcal{U}}$ corresponding to both Lagrangian functionals $\mathcal{L}_X(\hat{\rho}, \phi, \varphi)$ are indeed different, and therefore, Eq. (63) needs to be solved twice, one for each case, namely for $X = \eta$ and for $X = \tau$. Finally, the Finite Element implementation of both adjoint problems is shown Appendix B.

3.3.1. Descend direction for \mathcal{J}_{η}

For the case of the efficiency-based functional $\mathcal{J}_{\eta}(\hat{\rho}, \phi, \varphi)$ in (48), its directional derivative with respect to $\Delta \rho$, namely $D\mathcal{J}_{\eta}(\hat{\rho}, \phi, \varphi)[D\hat{\rho}[\Delta \rho]]$, necessary to compute the descend direction of the associated Lagrangian $\mathcal{L}_{\eta}(\rho, \phi, \varphi)$, namely $D\mathcal{L}_{\eta}(\rho, \phi, \varphi)[\Delta \rho]$ (see Eq. (61)), is obtained as

$$D\mathcal{J}_{\eta}(\hat{\rho}, \boldsymbol{\phi}, \varphi)[D\hat{\rho}[\Delta\rho]] = -\int_{\mathcal{B}_{0}} \left(\frac{\partial_{\hat{\rho}} \,\bar{\Psi}_{\mathrm{m}}^{\mathrm{mac}}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F}) \,\bar{\Psi}_{\mathrm{em}}^{\mathrm{mac}}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F}, \boldsymbol{E}_{0})}{\left(\bar{\Psi}_{\mathrm{em}}^{\mathrm{mac}}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F}, \boldsymbol{E}_{0}) \right)^{2}} \right) D\hat{\rho}[\Delta\rho] dV + \int_{\mathcal{B}_{0}} \left(\frac{\bar{\Psi}_{\mathrm{m}}^{\mathrm{mac}}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F}, \boldsymbol{E}_{0})}{\left(\bar{\Psi}_{\mathrm{em}}^{\mathrm{mac}}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F}, \boldsymbol{E}_{0}) \right)^{2}} \right) D\hat{\rho}[\Delta\rho] dV.$$

$$(64)$$

Finally, the directional derivatives of $\mathcal{J}_{\eta}(\hat{\rho}, \phi, \varphi)$ with respect to $\delta \phi$ and $\delta \varphi$, featuring in the adjoint problem (see Eq. (63)), are given, after standard derivation, by the following expressions

$$\begin{split} D\mathcal{J}_{\eta}(\hat{\rho}, \boldsymbol{\phi}, \varphi)[\delta \boldsymbol{\phi}] &= -\int_{\mathcal{B}_{0}} \frac{\bar{\Psi}_{\mathrm{em}}^{\mathrm{mac}}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F}, \boldsymbol{E}_{0})}{\left(\bar{\Psi}_{\mathrm{em}}^{\mathrm{mac}}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F}, \boldsymbol{E}_{0})\right)^{2}} \Big(\partial_{\boldsymbol{F}} \bar{\Psi}_{\mathrm{m}}^{\mathrm{mac}}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F}) : \nabla_{0} \delta \boldsymbol{\phi} \Big) dV \\ &+ \int_{\mathcal{B}_{0}} \frac{\bar{\Psi}_{\mathrm{m}}^{\mathrm{mac}}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F}, \boldsymbol{E}_{0})}{\left(\bar{\Psi}_{\mathrm{em}}^{\mathrm{mac}}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F}, \boldsymbol{E}_{0})\right)^{2}} \Big(\partial_{\boldsymbol{F}} \bar{\Psi}_{\mathrm{em}}^{\mathrm{mac}}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F}, \boldsymbol{E}_{0}) : \nabla_{0} \delta \boldsymbol{\phi} \Big) dV; \end{split}$$

$$D\mathcal{J}_{\eta}(\hat{\rho}, \boldsymbol{\phi}, \varphi)[\delta\varphi] = \int_{\mathcal{B}_{0}} \frac{\bar{\Psi}_{em}^{mac}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F}, \boldsymbol{E}_{0})}{\left(\bar{\Psi}_{em}^{mac}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F}, \boldsymbol{E}_{0})\right)^{2}} \left(\partial_{\boldsymbol{E}_{0}} \bar{\Psi}_{m}^{mac}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F}) \cdot \nabla_{0} \delta\varphi\right) dV - \int_{\mathcal{B}_{0}} \frac{\bar{\Psi}_{m}^{mac}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F}, \boldsymbol{E}_{0})}{\left(\bar{\Psi}_{em}^{mac}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F}, \boldsymbol{E}_{0})\right)^{2}} \left(\partial_{\boldsymbol{E}_{0}} \bar{\Psi}_{em}^{mac}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F}, \boldsymbol{E}_{0}) \cdot \nabla_{0} \delta\varphi\right) dV.$$
(65)

3.3.2. Descend direction for \mathcal{J}_{τ}

For the case of the stress-based functional $\mathcal{J}_{\tau}(\hat{\rho}, \phi, \varphi)$ in (49), its directional derivative with respect to $\Delta \rho$, namely $D\mathcal{J}_{\tau}(\hat{\rho}, \phi, \varphi)[D\hat{\rho}[\Delta \rho]]$, necessary to compute the descend direction of the associated Lagrangian $\mathcal{L}_{\tau}(\rho, \phi, \varphi)$, namely $D\mathcal{L}_{\tau}(\rho, \phi, \varphi)[\Delta \rho]$ (see Eq. (61)), is obtained as

$$D\mathcal{J}_{\tau}(\hat{\rho}, \boldsymbol{\phi})[D\hat{\rho}[\Delta\rho]] = (\mathcal{J}_{\tau}(\rho, \boldsymbol{\phi}))^{1-\gamma} \sum_{e}^{N} \left(\frac{\int_{\mathcal{B}_{0}^{e}} \tau(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F}) dV}{\tau_{\max} \int_{\mathcal{B}_{0}^{e}} dV} \right)^{1-\gamma} \int_{\mathcal{B}_{0}^{e}} D\tau[D\hat{\rho}[\Delta\rho]] dV;$$

$$D\tau[D\hat{\rho}[\Delta\rho]] = \frac{3}{\tau \hat{\rho}(\boldsymbol{X}, \boldsymbol{F})} \tau^{\text{dev}} : \mathbb{P}^{\text{dev}} : \partial_{\hat{\rho}} \left(\partial_{\boldsymbol{F}} \bar{\Psi}_{m}^{\text{mac}} \right) \boldsymbol{F}^{T}.$$
(66)

Finally, the directional derivative of $\mathcal{J}_{\tau}(\hat{\rho}, \phi, \varphi)$ with respect to $\delta \phi$, featuring in the adjoint problem (see Eq. (63)), is given, after standard derivation, by the following expression

$$D\mathcal{J}_{\tau}(\hat{\rho}, \boldsymbol{\phi})[\delta\boldsymbol{\phi}] = (\mathcal{J}_{\tau}(\rho, \boldsymbol{\phi}))^{1-\gamma} \sum_{e}^{N} \left(\frac{\int_{\mathcal{B}_{0}^{e}} \tau(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F}) dV}{\tau_{\max} \int_{\mathcal{B}_{0}^{e}} dV} \right)^{1-\gamma} \int_{\mathcal{B}_{0}^{e}} D\tau[\delta\boldsymbol{\phi}] dV;$$

$$D\tau[\delta\boldsymbol{\phi}] = \frac{3}{\tau\hat{\rho}(\boldsymbol{X}, \boldsymbol{F})} \tau^{\text{dev}} : \mathbb{P}^{\text{dev}} : \left(\partial_{\boldsymbol{F}\boldsymbol{F}}^{2} \bar{\Psi}_{m}^{\text{mac}} : \nabla_{0} \delta\boldsymbol{\phi} \right) \boldsymbol{F}^{T},$$

$$(67)$$

where it should be noted that in this case $D\mathcal{J}_{\tau}(\hat{\rho}, \phi)[\delta\varphi] = 0$, since the functional $\mathcal{J}_{\tau}(\hat{\rho}, \phi)$ does not depend on the field φ , unlike in the case of the functional $\mathcal{J}_{\eta}(\hat{\rho}, \phi, \varphi)$ (see Eq. (65)_b).

3.3.3. Algorithmic implementation

Finally, the reader interested in the algorithmic details of the optimisation algorithm implemented in our inhouse platform, can check the main ingredients necessary for computational implementation purposes in Algorithm 1, presented in a pseudo-code format.

4. Numerical examples

The objective of this section is to reveal the performance of the density-based topology optimisation approach described throughout Section 3. Specifically, the goal is to verify the capability of the proposed approach in the context of the Flexoelectric Micromorphic Elasticity (FM-Elasticity) formulation described through Section 2.

Prior to that, we describe the common features of all the numerical examples contained in the forthcoming subsections. First, we comment on the choice of the various penalising exponents $\{p_m, p_e, p_{mic}, p_{scale}, p_{flexo}\}$ featuring in the proposed interpolation scheme in Eq. (47). The reader can observe the values for these in Table 1. Notice from that table that $p_m = p_e = 3$. This choice is motivated by previous experience from the authors in the field of topology optimisation in nonlinear continuum electromechanics [72]. The choice of the other two parameters, namely $\{p_{mic}, p_{scale}\}$, is motivated by our empirical observations. With regard to the penalising scheme in the flexoelectric energy contribution, namely p_{flexo} , this deserves a careful discussion. After a detailed empirical analysis (i.e. numerical simulations), we observed that a considerably larger value was needed in comparison to the other four penalising exponents. The underlying reason for this difference in values lies in the fact that high strain gradients tend to occur in areas characterised by intermediate and low density values, generating a fictitious flexoelectric effect. In order to avoid this artificial effect, we advocate for higher values of the flexo-penalising exponent, hence, capturing flexoelectricity only in the purely solid regions and neglecting those associated with low density regions.

With regard to the Finite Element discretisation chosen, tri or (bi-)quadratic hexahedral Finite Element interpolations of the unknown fields $\{\phi, \varphi\}$ and tri or bi-linear for the field **G** have been considered in all the examples Set design variables ρ complying with volume constraint.

Get filtered density $\tilde{\rho}$ from (39) or (40).

Get physical density $\hat{\rho}$ from (41).

while user defined criterion do

Nonlinear forward problem in (62) and Appendix A. Get solution fields $\mathcal{U} = \{\phi, \varphi, G\}$.

Efficiency-based descend direction

- Linear adjoint problem in equations (63), (65) and Appendix B for functional $\mathcal{J}_{\eta}(\hat{\rho}, \phi, \varphi)$. Get adjoint fields $p_{\mathcal{U}} = \{p_{\phi}, p_{\varphi}, p_{G}\}$.
- Get descend direction $D\mathcal{L}_{\eta}[\Delta \phi]$ in (61) and (64).
- Evaluate functional \mathcal{J}_{η} in (48).

Stress-based descend direction

- Linear adjoint problem in equations (63), (67) and Appendix B for functional $\mathcal{J}_{\tau}(\hat{\rho}, \phi, \varphi)$. Get adjoint fields $p_{\mathcal{U}} = \{p_{\phi}, p_{\varphi}, p_{G}\}$.
- Get descend direction $D\mathcal{L}_{\tau}[\Delta \phi]$ in (61) and (66).
- Evaluate functional \mathcal{J}_{τ} in (49).

Evaluate objective function \mathcal{J} in terms of individual contributions \mathcal{J}_{η} and \mathcal{J}_{τ} according to (55).

Evaluate descend direction $D\mathcal{L}[\Delta\rho]$ in terms of individual contributions $D\mathcal{L}_{\eta}[\Delta\rho]$ and $D\mathcal{L}_{\tau}[\Delta\rho]$ according to (59).

Evaluate Volume constraint $g_{\Omega_0}(\rho)$, and derivative of volume constraint $Dg_{\Omega_0}(\rho)[\Delta\rho]$.

Pass { \mathcal{J} , $D\mathcal{L}[\Delta\phi]$, $g_{\mathcal{B}_0}(\rho)$, $Dg_{\mathcal{B}_0}(\rho)[\Delta\rho]$ } to optimisation code (**MMA**, **fmincon**, etc.) in order to **update** ρ .

end

Algorithm	1:	Optimisation	algorithm.
·			

Table 1 Numerical ex	amples. Penalisin	g exponents in the i	nterpolation scheme	e in Eq. (47).
Penalising e	xponents in the in	nterpolation scheme		
p _m	$p_{\rm e}$	$p_{ m mic}$	$p_{\rm scale}$	$p_{\rm flexo}$
3	3	3	3	9

(see Fig. 3), for 3-dimensional or 2-dimensional cases, respectively. Although a theoretical or numerical proof of the stability or inf-sup condition [89] associated with the aforementioned discretisation is not provided, our choice of Finite Element spaces is inspired by that in Ref. [77] in the linear regime. The authors in [77] did prove the stability of their discretisation by performing a numerical test following the work of Chapelle and Bathe [90]. Our discretisation is similar to that in [77], with the difference that we advocate for FM-Elasticity, whilst the formulation in [77] lies within FG-Elasticity. Therefore, the latter requires an extra degree of freedom in order to enforce (in a weak sense) the equivalence between the micro- and macro-deformation gradient tensors (i.e. G = F).

As with regard to the material parameters of the constitutive model, these are very similar to those considered in Ref. [44]. The various material parameters featuring in the different energetic contributions of the free energy density, described in Sections 2.5.1–2.5.4 can be observed in Table 2.



Fig. 3. Finite Element used in all numerical simulations for the two-dimensional case.

Table 2

Numerical examples. Material parameters featuring in the various energy contributions described through Sections 2.5.1–2.5.4.

$\mu_1 \text{ (N mm}^{-1})$	$\mu_2 \ (N \ mm^{-1})$	$\lambda \ (N \ mm^{-1})$	ε_r	$\varepsilon_0 (N kV^2)$	κ (N mm ⁻¹)	$\bar{\nu}_f$
9.84×10^{-3}	9.84×10^{-3}	30×10^{-3}	4.8	8.8542	$\mu_1 + \mu_2$	1
		0.1L				
		$\bigvee \bigvee \bigvee t = 1$	5×10^{-5}	$(N \cdot mm^{-1})$		
					.	
					Electrodes	
$\triangleleft _{1}$						
					х х	
V T					.	
V		L = 4(mm)	n)			
	<					

Fig. 4. Numerical Example 1. Geometry and boundary conditions.

With regard to the length scale parameter l featuring in the micromorphic contribution of the energy density in (28) and on the scaled flexoelectric coefficient in (31), this has been taken corresponding to three times the size of the representative size of the smallest Finite Element on each of the examples. Finally, the level of discretisation of the underlying Finite Element meshes used on each example is merely motivated by the degree of resolution sought for the density field. Relatively fine Finite Element discretisations, which permit the development of fine structural elements through the topology optimisation algorithm, are in general advocated for throughout the subsequent numerical examples.

4.1. Numerical example 1: bending induced flexoelectricity

The objective of this example is to demonstrate the applicability of the proposed formulation in a bending dominated problem. The geometry and boundary conditions of the problem can be seen in Fig. 4. The Finite Element discretisation comprises {161202, 80601, 81204} degrees of freedom for the unknown fields { ϕ , φ , G}. The volume fraction for this example (volume constraint in Eq. (56)) was set to f = 0.4.

With regard to the electrodes depicted in 4, notice that a prescribed value of $\varphi = 0$ is used for the grounded electrode (bottom electrode in Fig. 4). However, the other electrode does not have a prescribed value. Instead, the condition that needs to be prescribed is that the electric potential needs to be constant on the region (line) representing the electrode. This can be done in a variety of ways. In this work, we enforce it following a penalty-type formulation, where, for all the nodes $i = \{1, 2, ..., \alpha, ..., N\}$ associated with the discretisation of φ and



Fig. 5. Numerical Example 1. Optimal designs corresponding with various values of the weighting coefficient ω in Eq. (55), i.e.: (a) $\omega = 0$; (b) $\omega = 2$; (c) $\omega = 10$; (d) $\omega = 30$. Values of \mathcal{J}_{η} and \mathcal{J}_{τ} scaled with those of the case for $\omega = 0$.

contained on the electrode of interest, we enforce that all the values of φ at those nodes, namely φ_i , should be as close as possible to that of node α contained in the electrode. This can be done mathematically by modifying the Lagrangian in (57) as

$$\begin{split} \mathcal{L}(\rho, \mathcal{U}, \boldsymbol{p}_{\mathcal{U}}) &= \mathcal{J}(\phi, \phi, \varphi) - \left(D\Pi(\rho, \mathcal{U})[\boldsymbol{p}_{\phi}] + D\Pi(\rho, \mathcal{U})[\boldsymbol{p}_{\varphi}] + D\Pi(\rho, \mathcal{U})[\boldsymbol{p}_{G}] \right) \\ &- D\left(\sum_{i=1, i \neq \alpha}^{N} \frac{\tilde{\kappa}}{2} \left(\varphi_{i} - \varphi_{\alpha} \right)^{2} \right) [\boldsymbol{p}_{\varphi}] \\ &= \mathcal{J}(\phi, \phi, \varphi) - \left(D\Pi(\rho, \mathcal{U})[\boldsymbol{p}_{\phi}] + D\Pi(\rho, \mathcal{U})[\boldsymbol{p}_{\varphi}] + D\Pi(\rho, \mathcal{U})[\boldsymbol{p}_{G}] \right) \\ &- \sum_{i=1, i \neq \alpha}^{N} \tilde{\kappa} \left(\varphi_{i} - \varphi_{\alpha} \right) \boldsymbol{p}_{\varphi}, \end{split}$$

where $\tilde{\kappa} \in \mathbb{R}$ should have a sufficiently large value as to achieve the desired goal of enforcing that all the nodes have almost the same value of electric potential.

The optimal designs obtained can be observed in Fig. 5, for various values of the weighting coefficient ω in Eq. (55). Clearly, when $\omega = 0$, only the efficiency functional \mathcal{J}_{η} is active, which entails that the optimisation algorithm promotes the development of hinges (extremely thin structural elements), characterised by high strain gradients which obviously, yield a pronounced flexoelectric effect. This can be clearly reflected in Fig. 6. In the first row of this figure, we can observe that the highest extreme values for both Von Mises stress τ in (50) and electric field E_0 correspond with the case $\omega = 0$. In addition, the deformed configuration corresponding with this case is considerably larger when compared with that associated with other values of ω . The designs corresponding with increasing values of ω can also be observed in Fig. 5. Clearly, for the lowest value of $\omega \neq 0$, namely $\omega = 0$, this design is very similar from a topological standpoint, to its counterpart with $\omega = 0$. As the value ω increases, the dissimilarity in the designs with respect to the case $\omega = 0$ become much more apparent. This increasing topological dissimilarity manifest in Fig. 6 in the descend in the extreme values for both Von Mises stress and electric field, as expected.

As observed from Fig. 6, increasing values of ω entail a reduction of localised effects. This is reflected also in the values of efficiency \mathcal{J}_{η} (48) and \mathcal{J}_{τ} (49) for the various values of ω , which are shown in Fig. 5.

It is worth mentioning that the performance of the optimised flexoelectric design might be subjected to several sources of uncertainty including aleatory uncertainties, e.g. manufacturing imperfections, unknown loading conditions and variations of the material properties, and epistemic uncertainties due to the lack of knowledge. As a result, the designed performance predicted under deterministic assumption might be degraded. This issue becomes



Fig. 6. Numerical Example 1. Contour plot distribution of Von Mises stress τ (50) and the norm of the Lagrangian electric field E_0 for optimal designs in Fig. 5, corresponding with various values of the weighting coefficient ω in Eq. (55).

especially important in the manufacturing of micro- and nano-structures, where spatially varying manufacturing errors may cause parts of the structure to become thinner or thicker than intended. In the same vein, spatial variations of the electromechanical properties might lead to deviation from the predicted electric field. Although the field of topology optimisation under uncertainty is out of the scope of the current research, it is worth mentioning here due to its importance from the manufacturing point of view. Several probabilistic [91] and non-probabilistic [92] formulations have been presented in the topology optimisation literature addressing aforementioned issues. However, to the best of the authors' knowledge, the problem of topology optimisation under uncertainty has not been addressed so far in the context of flexoelectricity.

4.2. Numerical example 2: compressed energy harvester

In this example, we pursue a similar analysis to that in the example in Section 4.1, particularised to the geometry and boundary conditions of the problem can be seen in Fig. 7. The Finite Element discretisation comprises {161202, 80601, 81204} degrees of freedom for the unknown fields { ϕ , φ , G}. The volume fraction for this example (volume constraint in Eq. (56)) was set to f = 0.4.

Unlike in the preceding one in Section 4.1, where the flexoelectric effect was induced in a bending dominated problem, this example studies the generation of flexoelectricity through compression (see Fig. 7). The optimal designs obtained can be observed in Fig. 8, for two values of the weighting coefficient ω in Eq. (55). Clearly,



Fig. 7. Numerical Example 2. Geometry and boundary conditions.



Fig. 8. Numerical Example 2. Optimal designs corresponding with various values of the weighting coefficient ω in Eq. (55). Values of \mathcal{J}_{η} and \mathcal{J}_{τ} scaled with those of the case for $\omega = 0$.

when $\omega = 0$, only the efficiency functional \mathcal{J}_{η} is active, which entails that the optimisation algorithm promotes the development of hinges (extremely thin structural elements), characterised by high strain gradients which obviously, yield a pronounced flexoelectric effect. In fact, an extreme rotation yielding interpenetration can be observed in the deformed configuration associated with the case $\omega = 0$ on the left column in Fig. 9.

Once the value of ω is increased, the extreme deformation observed in 9 is considerably decreased in Fig. 9 (right column), the latter being also characterised by the presence of smaller values of the electric field and Von Mises stress distribution.



Fig. 9. Numerical Example 2. Contour plot distribution of Von Mises stress τ (50) and the norm of the Lagrangian electric field E_0 for optimal designs in Fig. 8, corresponding with the two values of the weighting coefficient ω in Eq. (55) considered.

4.3. Numerical example 3: torsion-induced flexoelectricity

The geometry and boundary conditions of the problem considered in this example can be seen in Fig. 10. The Finite Element discretisation comprises {321602, 160801, 161604} degrees of freedom for the unknown fields $\{\phi, \varphi, G\}$. The volume fraction for this example (volume constraint in Eq. (56)) was set to f = 0.5. This particular example investigates the generation of flexoelectricity through torsional induced deformations.

In addition, for a fixed value of the weighting parameter ω in Eq. (55), the influence in the design of the length scale parameter *R* featuring in the Helmholtz or PDE filter in Eq. (40) is investigated. Specifically, two values of *R* have been considered, namely $R = 0.08R_1$ and $R = 0.027R_1$. The designs corresponding with the two values of *R* can be seen in Fig. 11, together with their respective values for functionals \mathcal{J}_{η} and \mathcal{J}_{τ} . Both designs show that, as it was expected, the case with smaller length scale parameter *R* exhibits smaller design details, specially in the four diagonal elements and their unions with the fixed regions in the centre of the design.



Fig. 10. Numerical Example 3. Geometry and boundary conditions. The green area is fully clamped (fixed displacements). The value of the clockwise tangential force per unit length applied on the external circumference is $t = 6.36 \times 10^{-6}$ N mm⁻¹. (For interpretation of the references to colour in this figure legend, the reader is referred to the web version of this article.)



Fig. 11. Numerical Example 3. Optimal designs corresponding two different values in the length scale parameter R in the Helmholtz or PDE filter in Eq. (40) for a fixed value of $\omega = 2$: (a) $R = 0.08R_1$; (b) $R = 0.027R_1$. Values of \mathcal{J}_{η} and \mathcal{J}_{τ} scaled with those of the case for $R = 0.08R_1$.

Finally, the Von Mises stress distribution and the norm of the electric field for both designs in Fig. 11 are shown in Fig. 12.

4.4. Numerical example 4: 3D compressed flexoelectric cube

The final example explores the Flexo Micromorphic Finite Element formulation embedded into our in-house platform into a three-dimensional application. The geometry and boundary conditions are depicted and described in Fig. 13. The Finite Element discretisation comprises {985527, 85750, 385875} degrees of freedom for the unknown fields { ϕ , φ , G}, corresponding with one quarter of the domain (symmetry boundary conditions have been applied).



Fig. 12. Numerical Example 3. Contour plot distribution of Von Mises stress τ (50) and the norm of the Lagrangian electric field E_0 for optimal designs in Fig. 11, corresponding with the two values of the length scale parameter R in the Helmholtz or PDE filter in Eq. (40) and for a fixed value of weighting coefficient ω in Eq. (55).

For the configuration described, Figs. 14a and b show the designs for a volume fraction of f = 0.45 for different values of the weighting coefficient ω in Eq. (55). The difference between both designs becomes more apparent in Fig. 15, where their respective contour plot distribution of Von Mises stress and electric field is shown. As expected, the design with $\omega = 0.3$ yields smaller extreme values as its counterpart with $\omega = 0.1$. In addition, the design and Von Mises stress and electric field distribution for the case of smaller volume fraction (i.e. f = 0.3) with $\omega = 0.3$ is shown in Figs. 14 and 15 too.

5. Conclusions

This paper presents the first work where topology optimisation is applied for the design of dielectric elastomerbased flexoelectric energy harvester, hence, in the context of finite strains. The micromorphic continuum formulation described in Ref. [44] has been adopted for the modelling of flexoelectricity, in conjunction with a mixed-type Finite Element formulation, permitting the use of C^0 approximations for the different fields involved, namely, displacements, electric potential and the micro deformation gradient tensor.

From the topology optimisation standpoint, a novel energy interpolation scheme has been presented, whereby different penalising exponents are considered for the various contributions that the free energy function is additively



Fig. 13. Numerical Example 4. Geometry and boundary conditions for cube of side 1 mm. The two internal planes, in addition to the six lateral faces are connected to ground ($\varphi = 0$). Fixed displacements are imposed in the red region in the bottom lateral face. The value of the compressive force per unit area in the top face of the cube is $t = 10^{-6}$ N mm⁻¹. (For interpretation of the references to colour in this figure legend, the reader is referred to the web version of this article.)

decomposed into. Crucially, and motivated by purely empirical observations, this paper proposes the use of higher values of the penalising scheme in the flexoelectric energy contribution. The underlying reason for this difference in values lies in the fact that high strain gradients tend to occur in areas characterised by intermediate and low density values, generating a fictitious flexoelectric effect. In order to avoid this artificial effect, we advocate for higher values of the flexo-penalising exponent, hence, capturing flexoelectricity only in the purely solid regions and neglecting those associated with low density regions.

A multi-objective type formulation has been proposed whereby a weighted combination of efficiency-based measures and aggregation functions of the Von Mises stress distribution within the material are considered. This was motivated by the necessity to remedy some shortcomings of the efficiency-based functional, promoting the development of hinges with highly localised large strain gradients, hence, yielding designs which are inappropriate from a structural standpoint. The effect of the consideration of the stress-based functional has been evidenced in the various numerical examples shown in Section 4. These have shown that an increasing contribution of the latter function alleviates and even eliminates the presence of hinges, which in turn, manifest in an obvious decrease of the electrical efficiency, but with the benefit of yielding designs with a better structural performance.

An important physical phenomenon inherent in dielectric-based flexoelectric devices is the possible development of electrical breakdown [88], occurring when the electric field exceeds the dielectric strength of the dielectric material. Consideration of electrical breakdown in the design of dielectric devices and flexoelectric devices, to the best of the authors' knowledge, has never been incorporated in the design problem, despite its importance. In addition, the value of the external mechanical forces in all the numerical examples on this paper has led to moderate deformations, except in the cases where a vanishing weighting factor associated with the aggregation function of the Von Mises stress distribution was considered. Although the latter parameter, up to a certain extent, helps preventing the development of mechanically induced instabilities, consideration of buckling-type constraints in the optimisation problem can also be investigated as an alternative form of preventing localised deformations leading to poor structural performance.

In the numerical examples section, the consideration of a vanishing factor ω (i.e. negligible influence of the Von Mises type objective function) yields extremely thin structural elements or hinges, which are promoted by the optimisation process as an efficient way to create strain gradients and, hence, electric fields. The deformed



Fig. 14. Numerical Example 4. Optimal designs with different volume fractions f and weighting parameters ω in Eq. (55).

configurations associated with these figures show very large deformations, yielding rotations around the hinges of the design. For the cases when ω is different from zero, hinges are precluded. Nonetheless, the optimisation process promotes an alternative mechanism which favours the development of strain gradients through the creation of structural curved elements. For the values of the external mechanical loads considered in this paper, relatively moderate deformations have been obtained in these cases. Therefore, mechanically induced instabilities have not been observed. For larger values of applied forces, the global stability of the proposed design might be indeed compromised. For that case, consideration of buckling-type constraints could be incorporated with the aim of increasing the stability of the design.



Fig. 15. Numerical Example 4. Contour plot distribution of Von Mises stress τ (50) and the norm of the Lagrangian electric field E_0 for optimal designs in Fig. 14.

Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Data availability

Data will be made available on request.

Acknowledgements

The first author acknowledges the financial support through the contract 21132/SF/19, Fundación Séneca, Región de Murcia (Spain), through the program Saavedra Fajardo. First and second authors are funded by Fundación Séneca (Murcia, Spain) through grant 20911/PI/18. First and second authors also acknowledge the financial support through grant PID2021-125687OA-I00. The third author acknowledges the financial support received through the European Training Network Protection (Project ID: 764636).

Compliance with ethical standards

Replication of results Regarding reproducibility of results, the following remarks apply:

- The code used to run the numerical examples in Section 4 is comprised of the following ingredients: (i) MMA algorithm [93]; (ii) 3D SIMP code [94]; (iii) the finite element formulation for the flexoelectric continuum formulation, described in Appendix A (for the forward problem) and Appendix B, for the two adjoint problems.
- For those readers interested in using our in-house platform, the authors are happy to be contacted.
- Algorithm 1 presents in pseudo-code format, the steps required and main ingredients necessary for the computational implementation of the optimisation algorithm followed in this paper.

Appendix A. Finite element implementation of forward problem

As standard in finite elements, the domain \mathcal{B}_0 described in Section 2.1, is sub-divided into a finite set of non-overlapping elements $e \in \mathbb{E}$ as follows

$$\mathcal{B}_0 \approx \mathcal{B}_0^h = \bigcup_{e \in \mathbb{E}} \mathcal{B}_0^e.$$
(A.1)

The unknown fields $\{\phi, \varphi, G\}$, and the test functions $\{p_{\phi}, p_{\varphi}, p_{G}\}$ featuring in $D\Pi[p_{\phi}]$, $D\Pi[p_{\varphi}]$ and $D\Pi[p_{G}]$ in (13), are discretised using the functional spaces $\mathbb{V}^{\phi^{h}} \times \mathbb{V}^{\varphi^{h}} \times \mathbb{V}^{G^{h}}$ and $\mathbb{V}_{0}^{\phi^{h}} \times \mathbb{V}_{0}^{G^{h}}$, respectively, defined as

$$\mathbb{V}^{\boldsymbol{\phi}^{h}} = \left\{ \boldsymbol{\phi} \in \mathbb{V}^{\boldsymbol{\phi}}; \quad \boldsymbol{\phi}^{h} \big|_{\mathcal{B}^{e}_{0}} = \sum_{a=1}^{n_{\text{node}}^{\boldsymbol{\phi}}} N_{a}^{\boldsymbol{\phi}} \boldsymbol{\phi}_{a} \right\}; \qquad \mathbb{V}^{\boldsymbol{\phi}^{h}} = \left\{ \boldsymbol{\varphi} \in \mathbb{V}^{\boldsymbol{\varphi}}; \quad \boldsymbol{\varphi}^{h} \big|_{\mathcal{B}^{e}_{0}} = \sum_{a=1}^{n_{\text{node}}^{\boldsymbol{\varphi}}} N_{a}^{\boldsymbol{\varphi}} \boldsymbol{\varphi}_{a} \right\};$$

$$\mathbb{V}^{\boldsymbol{G}^{h}} = \left\{ \boldsymbol{G} \in \mathbb{V}^{\boldsymbol{G}}; \quad \boldsymbol{G}^{h} \big|_{\mathcal{B}^{e}_{0}} = \sum_{a=1}^{n_{\text{node}}^{\boldsymbol{G}}} N_{a}^{\boldsymbol{G}} \boldsymbol{G}_{a} \right\}$$
(A.2)

where, for any field $\mathcal{U} \in \{\phi, \varphi, G\}$, $n_{node}^{\mathcal{U}}$ denotes the number of nodes per element of the discretisation associated with the field \mathcal{U} ; and $N_a^{\mathcal{U}} : \mathcal{B}_0^e \to \mathbb{R}$, the *a*th shape function used for the interpolation of \mathcal{U} . In addition, \mathcal{U}_a represents the value of the field \mathcal{U} at the *a*th node of a given finite element. Consideration of the functional spaces in (A.2) enables the weak forms $D\Pi[p_{\varphi}]$, $D\Pi[p_{\varphi}]$ and $D\Pi[p_G]$ in (13) to be written in terms of their associated elemental residual contributions, namely

$$D\Pi[\boldsymbol{p}_{\boldsymbol{\phi}}] = \sum_{e=1}^{N} \boldsymbol{p}_{\boldsymbol{\phi}_{a}} \cdot \boldsymbol{R}_{a,e}^{\boldsymbol{\phi}}; \qquad D\Pi[\boldsymbol{p}_{\varphi}] = \sum_{e=1}^{N} \boldsymbol{p}_{\varphi_{a}} \boldsymbol{R}_{a,e}^{\varphi}; \qquad D\Pi[\boldsymbol{p}_{G}] = \sum_{e=1}^{N} \boldsymbol{p}_{Ga} \cdot \boldsymbol{R}_{a,e}^{G}, \tag{A.3}$$

where N denotes the number of elements for the underlying discretisation, and where each of the residual contributions $\mathbf{R}_{a,e}^{\phi}$, $\mathbf{R}_{a,e}^{\varphi}$ and $\mathbf{R}_{a,e}^{G}$ can be expressed as²

$$\begin{aligned} \boldsymbol{R}_{a,e}^{\boldsymbol{\phi}} &= \int_{\mathcal{B}_{0}^{e}} \left(\partial_{F} \bar{\Psi} \left(\hat{\rho}(\boldsymbol{X}), \mathcal{F} \right) \right) \nabla_{0} N_{a}^{\boldsymbol{\phi}} \, dV + \int_{\mathcal{B}_{0}^{e}} N_{a}^{\boldsymbol{\phi}} \boldsymbol{f}_{0} \, dV; \\ \boldsymbol{R}_{a,e}^{\varphi} &= -\int_{\mathcal{B}_{0}^{e}} \left(\partial_{E_{0}} \bar{\Psi} \left(\hat{\rho}(\boldsymbol{X}), \mathcal{F} \right) \right) \cdot \nabla_{0} N_{a}^{\varphi} \, dV + \int_{\mathcal{B}_{0}^{e}} N_{a}^{\varphi} \rho^{e} \, dV; \\ \boldsymbol{R}_{a,e}^{\boldsymbol{G}} &= \int_{\mathcal{B}_{0}} \left(\partial_{\boldsymbol{G}} \bar{\Psi} \left(\hat{\rho}(\boldsymbol{X}), \mathcal{F} \right) \right) N_{a}^{\boldsymbol{G}} \, dV + \int_{\mathcal{B}_{0}} \left(\partial_{\boldsymbol{\mathcal{G}}} \bar{\Psi} \left(\hat{\rho}(\boldsymbol{X}), \mathcal{F} \right) \right) \cdot \nabla_{0} N_{a}^{\boldsymbol{G}} \, dV. \end{aligned}$$
(A.4)

Discretisation of the test functions and incremental fields permits Eq. (17) to be written in terms of their associated elemental stiffness contributions, namely

$$D^{2}\Pi[\boldsymbol{p}_{\phi}; \Delta \boldsymbol{\phi}] = \sum_{e=1}^{N} \boldsymbol{p}_{\phi_{a}} \cdot \boldsymbol{K}_{ab,e}^{\phi\phi} \Delta \boldsymbol{\phi}_{b}; \qquad D^{2}\Pi[\boldsymbol{p}_{\phi}; \Delta \varphi] = \sum_{e=1}^{N} \boldsymbol{p}_{\phi_{a}} \cdot \boldsymbol{K}_{ab,e}^{\phi\varphi} \Delta \varphi_{b};$$

$$D^{2}\Pi[\boldsymbol{p}_{\phi}; \Delta \boldsymbol{G}] = \sum_{e=1}^{N} \boldsymbol{p}_{\phi_{a}} \cdot \boldsymbol{K}_{ab,e}^{\phi\phi} \Delta \boldsymbol{\phi}_{b}; \qquad D^{2}\Pi[\boldsymbol{p}_{\phi}; \Delta \varphi] = \sum_{e=1}^{N} \boldsymbol{p}_{\varphi_{a}} \cdot \boldsymbol{K}_{ab,e}^{\phi\varphi} \Delta \varphi_{b};$$

$$D^{2}\Pi[\boldsymbol{p}_{\phi}; \Delta \boldsymbol{G}] = \sum_{e=1}^{N} \boldsymbol{p}_{\varphi_{a}} \cdot \boldsymbol{K}_{ab,e}^{\phi \boldsymbol{G}} \Delta \boldsymbol{G}_{b};$$

$$D^{2}\Pi[\boldsymbol{p}_{\phi}; \Delta \boldsymbol{G}] = \sum_{e=1}^{N} \boldsymbol{p}_{\varphi_{a}} \cdot \boldsymbol{K}_{ab,e}^{\phi \boldsymbol{G}} \Delta \boldsymbol{G}_{b};$$

$$D^{2}\Pi[\boldsymbol{p}_{\phi}; \Delta \boldsymbol{G}] = \sum_{e=1}^{N} \boldsymbol{p}_{G_{a}} \cdot \boldsymbol{K}_{ab,e}^{\phi \boldsymbol{G}} \Delta \boldsymbol{G}_{b};$$

$$D^{2}\Pi[\boldsymbol{p}_{G}; \Delta \boldsymbol{\phi}] = \sum_{e=1}^{N} \boldsymbol{p}_{G_{a}} \cdot \boldsymbol{K}_{ab,e}^{G\phi} \Delta \boldsymbol{\phi}_{b}; \qquad D^{2}\Pi[\boldsymbol{p}_{G}; \Delta \varphi] = \sum_{e=1}^{N} \boldsymbol{p}_{G_{a}} \cdot \boldsymbol{K}_{ab,e}^{G\phi} \Delta \varphi_{b};$$

$$D^{2}\Pi[\boldsymbol{p}_{G}; \Delta \boldsymbol{G}] = \sum_{e=1}^{N} \boldsymbol{p}_{G_{a}} \cdot \boldsymbol{K}_{ab,e}^{G\phi} \Delta \boldsymbol{G}_{b},$$
(A.5)

where each of the stiffness contributions is expressed as

$$\begin{split} \left(\boldsymbol{K}_{ab,e}^{\boldsymbol{\phi}\boldsymbol{\phi}}\right)_{ij} &= \int_{\mathcal{B}_{0}^{e}} \left(\boldsymbol{\nabla}_{0}N_{a}^{\boldsymbol{\phi}}\right)_{I} \left(\boldsymbol{\nabla}_{0}N_{b}^{\boldsymbol{\phi}}\right)_{J} \left(\partial_{FF}^{2}\bar{\Psi}(\hat{\rho}(\boldsymbol{X}),\mathcal{F})\right)_{iIJJ} dV; \\ \left(\boldsymbol{K}_{ab,e}^{\boldsymbol{\phi}\boldsymbol{\phi}}\right)_{i} &= -\int_{\mathcal{B}_{0}^{e}} \left(\boldsymbol{\nabla}_{0}N_{a}^{\boldsymbol{\phi}}\right)_{I} \left(\boldsymbol{\nabla}_{0}N_{b}^{\boldsymbol{\phi}}\right)_{J} \left(\partial_{FE_{0}}^{2}\bar{\Psi}(\hat{\rho}(\boldsymbol{X}),\mathcal{F})\right)_{iIJJ} dV; \\ \left(\boldsymbol{K}_{ab,e}^{\boldsymbol{\phi}\boldsymbol{G}}\right)_{iJJ} &= \int_{\mathcal{B}_{0}^{e}} \left(\boldsymbol{\nabla}_{0}N_{a}^{\boldsymbol{\phi}}\right)_{I} N_{b}^{G} \left(\partial_{FG}^{2}\bar{\Psi}(\hat{\rho}(\boldsymbol{X}),\mathcal{F})\right)_{iIJJ} dV; \\ \left(\boldsymbol{K}_{ab,e}^{\varphi\boldsymbol{\phi}}\right) &= \int_{\mathcal{B}_{0}^{e}} \left(\boldsymbol{\nabla}_{0}N_{a}^{\varphi}\right)_{I} N_{b}^{G} \left(\partial_{E_{0}G}^{2}\bar{\Psi}(\hat{\rho}(\boldsymbol{X}),\mathcal{F})\right) \boldsymbol{\nabla}_{0}N_{b}^{\varphi} dV; \\ \left(\boldsymbol{K}_{ab,e}^{\varphi\boldsymbol{G}}\right)_{iI} &= -\int_{\mathcal{B}_{0}^{e}} \left(\boldsymbol{\nabla}_{0}N_{a}^{\varphi}\right)_{J} N_{b}^{G} \left(\partial_{E_{0}G}^{2}\bar{\Psi}(\hat{\rho}(\boldsymbol{X}),\mathcal{F})\right)_{JiI} dV \\ &- \int_{\mathcal{B}_{0}^{e}} \left(\boldsymbol{\nabla}_{0}N_{a}^{\varphi}\right)_{J} \left(\boldsymbol{\nabla}_{0}N_{b}^{G}\right)_{K} \left(\partial_{E_{0}G}^{2}\bar{\Psi}(\hat{\rho}(\boldsymbol{X}),\mathcal{F})\right)_{JiIK} dV; \end{split}$$

 $^{^{2}}$ For simplicity, the external contributions on the boundary of the continuum have not been included in (A.4).

$$\begin{split} \left(\boldsymbol{K}_{ab,e}^{\boldsymbol{G}\boldsymbol{G}}\right)_{iIjJ} &= \int_{\mathcal{B}_{0}^{e}} N_{a}^{\boldsymbol{G}} N_{b}^{\boldsymbol{G}} \left(\partial_{\boldsymbol{G}\boldsymbol{G}}^{2} \bar{\boldsymbol{\Psi}}(\hat{\boldsymbol{\rho}}(\boldsymbol{X}), \boldsymbol{\mathcal{F}})\right)_{iIjJ} dV + \int_{\mathcal{B}_{0}^{e}} N_{a}^{\boldsymbol{G}} \left(\boldsymbol{\nabla}_{0} N_{b}^{\boldsymbol{G}}\right)_{K} \left(\partial_{\boldsymbol{G}\boldsymbol{G}\boldsymbol{G}}^{2} \bar{\boldsymbol{\Psi}}(\hat{\boldsymbol{\rho}}(\boldsymbol{X}), \boldsymbol{\mathcal{F}})\right)_{iIjJK} dV; \\ &+ \int_{\mathcal{B}_{0}^{e}} \left(\boldsymbol{\nabla}_{0} N_{a}^{\boldsymbol{G}}\right)_{K} N_{b}^{\boldsymbol{G}} \left(\partial_{\boldsymbol{G}\boldsymbol{G}}^{2} \bar{\boldsymbol{\Psi}}(\hat{\boldsymbol{\rho}}(\boldsymbol{X}), \boldsymbol{\mathcal{F}})\right)_{iIKjJ} dV \\ &+ \int_{\mathcal{B}_{0}^{e}} \left(\boldsymbol{\nabla}_{0} N_{a}^{\boldsymbol{G}}\right)_{K} \left(\boldsymbol{\nabla}_{0} N_{b}^{\boldsymbol{G}}\right)_{L} \left(\partial_{\boldsymbol{G}}^{2} \boldsymbol{\mathcal{G}} \bar{\boldsymbol{\Psi}}(\hat{\boldsymbol{\rho}}(\boldsymbol{X}), \boldsymbol{\mathcal{F}})\right)_{iIKjJL} dV, \end{split}$$
(A.6)

with the remaining three contributions $K_{ab,e}^{\varphi\phi}$, $K_{ab,e}^{G\phi}$ and $K_{ab,e}^{G\varphi}$ defined as

$$\boldsymbol{K}_{ab,e}^{\boldsymbol{\varphi}\boldsymbol{\phi}} = \left(\boldsymbol{K}_{ab,e}^{\boldsymbol{\varphi}\boldsymbol{\varphi}}\right)^{T}; \qquad \boldsymbol{K}_{ab,e}^{\boldsymbol{G}\boldsymbol{\phi}} = \left(\boldsymbol{K}_{ab,e}^{\boldsymbol{\varphi}\boldsymbol{G}}\right)^{T}; \qquad \boldsymbol{K}_{ab,e}^{\boldsymbol{G}\boldsymbol{\varphi}} = \left(\boldsymbol{K}_{ab,e}^{\boldsymbol{\varphi}\boldsymbol{G}}\right)^{T}.$$
(A.7)

Assembly over all the elements of the underlying discretisations leads to the following algebraic linear system of equations

$$\begin{bmatrix} K^{\phi\phi} & K^{\phi\varphi} & K^{\phi G} \\ K^{\phi\phi} & K^{\phi\varphi} & K^{\phi G} \\ K^{G\phi} & K^{G\varphi} & K^{GG} \end{bmatrix} \begin{bmatrix} \widetilde{\Delta\phi} \\ \widetilde{\Delta\phi} \\ \widetilde{\DeltaG} \end{bmatrix} = \begin{bmatrix} R^{\phi} \\ R^{\varphi} \\ R^{G} \end{bmatrix},$$
(A.8)

where K^{XY} refers to the assembled counterpart of the element contribution $K^{XY}_{ab,e}$, being X and Y any member of the set $\{\phi, \varphi, G\}$. In addition, in Eq. (A.8), R^X represents the assembled residual vector associated with the element contribution $R^X_{a,e}$. Finally, $\widetilde{\Delta\phi}$, $\widetilde{\Delta\varphi}$ and $\widetilde{\Delta G}$ represent the vector of nodal values associated with the unknown fields ϕ , φ and G, respectively.

Appendix B. Finite element implementation of adjoint problem

We recast Eq. (B.1), where the adjoint problems (recall that there is one adjoint problem associated with the efficiency-based functional \mathcal{J}_{η} (48) and another associated with the stress-based functional \mathcal{J}_{τ} (49)) can be written in the following compact manner

(B.1)

with X referring to the type of functional considered, i.e. $X = \eta$, τ . Taking $\{\delta \phi, \delta \varphi, \delta G\} \in \mathbb{V}_0^{\phi^h} \times \mathbb{V}_0^{\phi^h} \times \mathbb{V}_0^{G^h}$ and $\{p_{\phi}, p_{\varphi}, p_G\} \in \mathbb{V}_0^{\phi^h} \times \mathbb{V}_0^{G^h}$, discretisation of (B.1) permits to rewrite the derivatives of the functionals \mathcal{J}_X therein as

$$D\mathcal{J}_X[\delta \boldsymbol{\phi}] = \sum_{e=1}^N \delta \boldsymbol{\phi}_a \cdot \boldsymbol{R}^{\boldsymbol{\phi}}_{X_{a,e}}; \qquad D\mathcal{J}_X[\delta \varphi] = \sum_{e=1}^N p_{\varphi_a} R^{\varphi}_{a,e}; \qquad D\mathcal{J}_X[\delta \boldsymbol{G}] = 0.$$
(B.2)

where, for the case of the efficiency-based functional \mathcal{J}_{η} , the elemental residual contributions are

$$\begin{split} \boldsymbol{R}_{\eta_{a,e}}^{\boldsymbol{\phi}} &= -\int_{\mathcal{B}_{0}} \frac{\bar{\Psi}_{\mathrm{em}}^{\mathrm{mac}}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F}, \boldsymbol{E}_{0})}{\left(\bar{\Psi}_{\mathrm{em}}^{\mathrm{mac}}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F}, \boldsymbol{E}_{0})\right)^{2}} \left(\left(\partial_{F} \bar{\Psi}_{\mathrm{m}}^{\mathrm{mac}}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F})\right) \nabla_{0} N_{a}^{\boldsymbol{\phi}} \right) dV \\ &+ \int_{\mathcal{B}_{0}} \frac{\bar{\Psi}_{\mathrm{m}}^{\mathrm{mac}}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F})}{\left(\bar{\Psi}_{\mathrm{em}}^{\mathrm{mac}}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F}, \boldsymbol{E}_{0})\right)^{2}} \left(\left(\partial_{F} \bar{\Psi}_{\mathrm{em}}^{\mathrm{mac}}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F}, \boldsymbol{E}_{0})\right) \nabla_{0} N_{a}^{\boldsymbol{\phi}} \right) dV; \end{split}$$

$$\boldsymbol{R}_{\eta_{a,e}}^{\varphi} = \int_{\mathcal{B}_{0}} \frac{\bar{\Psi}_{em}^{mac}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F}, \boldsymbol{E}_{0})}{\left(\bar{\Psi}_{em}^{mac}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F}, \boldsymbol{E}_{0})\right)^{2}} \left(\left(\partial_{\boldsymbol{E}_{0}} \bar{\Psi}_{m}^{mac}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F})\right) \nabla_{0} N_{a}^{\varphi} \right) dV
- \int_{\mathcal{B}_{0}} \frac{\bar{\Psi}_{m}^{mac}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F})}{\left(\bar{\Psi}_{em}^{mac}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F}, \boldsymbol{E}_{0})\right)^{2}} \left(\left(\partial_{\boldsymbol{E}_{0}} \bar{\Psi}_{em}^{mac}(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F}, \boldsymbol{E}_{0})\right) \nabla_{0} N_{a}^{\varphi} \right) dV.$$
(B.3)

On the other hand, for the case of the stress-based functional $\mathcal{J}_{\tau}(\hat{\rho}, \phi)$, we have

$$\begin{pmatrix} \boldsymbol{R}^{\boldsymbol{\phi}}_{\tau_{a,e}} \end{pmatrix}_{i} = (\mathcal{J}_{\tau}(\rho, \boldsymbol{\phi}))^{1-\gamma} \sum_{e}^{N} \left(\frac{\int_{\mathcal{B}^{e}_{0}} \tau(\hat{\rho}(\boldsymbol{X}), \boldsymbol{F}) dV}{\tau_{\max} \int_{\mathcal{B}^{e}_{0}} dV} \right)^{1-\gamma} \int_{\mathcal{B}^{e}_{0}} (\boldsymbol{r})_{i} dV;$$

$$(\boldsymbol{r})_{i} = \frac{3}{\tau \hat{\rho}(\boldsymbol{X}, \boldsymbol{F})} \left(\tau^{\text{dev}} : \mathbb{P}^{\text{dev}} \right)_{rs} \left(\partial_{FF}^{2} \bar{\Psi}^{\text{mac}}_{m} \nabla_{0} N^{\boldsymbol{\phi}}_{a} \right)_{rmi} (\boldsymbol{F})_{sm},$$

$$(B.4)$$

and $\mathbf{R}_{\tau_{a,e}}^{\varphi} = 0$. Similarly, upon discretisation, the different terms of the form $D^2 \Pi(\rho, \mathcal{U})[\delta \mathcal{U}; \mathcal{U}]$ lead to the following expressions

$$D^{2}\Pi[\delta\boldsymbol{\phi}; \boldsymbol{p}_{\boldsymbol{\phi}}] = \sum_{e=1}^{N} \delta\boldsymbol{\phi}_{a} \cdot \boldsymbol{K}_{ab,e}^{\boldsymbol{\phi}\boldsymbol{\phi}} \boldsymbol{p}_{\boldsymbol{\phi}_{b}}; \qquad D^{2}\Pi[\delta\boldsymbol{\phi}; \boldsymbol{p}_{\varphi}] = \sum_{e=1}^{N} \delta\boldsymbol{\phi}_{a} \cdot \boldsymbol{K}_{ab,e}^{\boldsymbol{\phi}\varphi} \boldsymbol{p}_{\varphi_{b}}; \\D^{2}\Pi[\delta\boldsymbol{\phi}; \boldsymbol{p}_{\boldsymbol{\sigma}}] = \sum_{e=1}^{N} \delta\boldsymbol{\phi}_{a} \cdot \boldsymbol{K}_{ab,e}^{\boldsymbol{\phi}\boldsymbol{\phi}} \boldsymbol{p}_{\boldsymbol{\phi}_{b}}; \qquad D^{2}\Pi[\delta\varphi; \boldsymbol{p}_{\varphi}] = \sum_{e=1}^{N} \delta\varphi_{a} \cdot \boldsymbol{K}_{ab,e}^{\varphi\varphi} \boldsymbol{p}_{\varphi_{b}}; \\D^{2}\Pi[\delta\varphi; \boldsymbol{p}_{\boldsymbol{\phi}}] = \sum_{e=1}^{N} \delta\varphi_{a} \cdot \boldsymbol{K}_{ab,e}^{\varphi\varphi} \boldsymbol{p}_{\boldsymbol{G}_{b}}; \qquad D^{2}\Pi[\delta\varphi; \boldsymbol{p}_{\varphi}] = \sum_{e=1}^{N} \delta\boldsymbol{G}_{a} \cdot \boldsymbol{K}_{ab,e}^{\varphi\varphi} \boldsymbol{p}_{\varphi_{b}}; \\D^{2}\Pi[\delta\boldsymbol{G}; \boldsymbol{p}_{\boldsymbol{\phi}}] = \sum_{e=1}^{N} \delta\boldsymbol{G}_{a} \cdot \boldsymbol{K}_{ab,e}^{\boldsymbol{G}\boldsymbol{\phi}} \boldsymbol{p}_{\boldsymbol{\phi}_{b}}; \qquad D^{2}\Pi[\delta\boldsymbol{G}; \boldsymbol{p}_{\varphi}] = \sum_{e=1}^{N} \delta\boldsymbol{G}_{a} \cdot \boldsymbol{K}_{ab,e}^{\boldsymbol{G}\varphi} \boldsymbol{p}_{\varphi_{b}}; \\D^{2}\Pi[\delta\boldsymbol{G}; \boldsymbol{p}_{\boldsymbol{\phi}}] = \sum_{e=1}^{N} \delta\boldsymbol{G}_{a} \cdot \boldsymbol{K}_{ab,e}^{\boldsymbol{G}\boldsymbol{\phi}} \boldsymbol{p}_{\boldsymbol{\phi}_{b}}, \qquad D^{2}\Pi[\delta\boldsymbol{G}; \boldsymbol{p}_{\varphi}] = \sum_{e=1}^{N} \delta\boldsymbol{G}_{a} \cdot \boldsymbol{K}_{ab,e}^{\boldsymbol{G}\varphi} \boldsymbol{p}_{\varphi_{b}};$$

where each of the stiffness contributions is exactly the same as the stiffness matrices arising from the linearisation of the forward problem (see Eq. (A.6)). Assembly over all the elements of the underlying discretisations leads to the following algebraic linear system of equations

$$\begin{bmatrix} K^{\phi\phi} & K^{\phi\varphi} & K^{\phi G} \\ K^{\phi\phi} & K^{\varphi\varphi} & K^{\varphi G} \\ K^{G\phi} & K^{G\varphi} & K^{GG} \end{bmatrix} \begin{bmatrix} \widetilde{p}_{\phi} \\ \widetilde{p}_{\phi} \\ \widetilde{p}_{G} \end{bmatrix} = \begin{bmatrix} R^{\phi}_{\chi} \\ R^{\phi}_{\chi} \\ \mathbf{0} \end{bmatrix},$$
(B.6)

where K^{XY} refers to the assembled counterpart of the element contribution $K_{ab,e}^{XY}$, being X and Y any member of the set $\{\phi, \varphi, G\}$. In addition, in Eq. (B.6), R^X represents the assembled residual vector associated with the element contribution $R_{X_{a,e}}^X$. Finally, $\widetilde{p_{\phi}}$, $\widetilde{p_{\varphi}}$ and $\widetilde{p_G}$ represent the vector of nodal values associated with the adjoint fields p_{ϕ} , p_{φ} and p_G , respectively.

References

- [1] A.K. Tagantsev, Piezoelectricity and flexoelectricity in crystalline dielectrics, Phys. Rev. B 34 (1986) 5883–5889.
- [2] P. Neff, A. Madeo, G. Barbagallo, M.V. D'Agostino, R. Abreu, I.-D. Ghiba, Real wave propagation in the isotropic-relaxed micromorphic model, Proc. R. Soc. Lond. Ser. A Math. Phys. Eng. Sci. 473 (2017).
- [3] A.G. Petrov, Electricity and mechanics of biomembrane systems: flexoelectricity in living membranes, Anal. Chim. Acta 568 (2006) 1–2.

- [4] R. Maranganti, P. Sharma, Atomistic determination of flexoelectric properties of crystalline dielectrics, Phys. Rev. B 80 (2009) 054109.
- [5] A. Abdollahi, I. Arias, Constructive and destructive interplay between piezoelectricity and flexoelectricity in flexural sensors and actuators, J. Appl. Mech. Trans. ASME 82 (2015) 121003–1–121003–4.
- [6] Pavlo Zubko, Gustau Catalan, Alexander K. Tagantsev, Flexoelectric effect in solids, Annu. Rev. Mater. Res. 43 (1) (2013) 387-421.
- [7] M.S. Majdoub, P. Sharma, T. Cagin, Enhanced size-dependent piezoelectricity and elasticity in nanostructures due to the flexoelectric effect, Phys. Rev. B 77 (2008) 125424.
- [8] Amir Abdollahi, Christian Peco, Daniel Millán, Marino Arroyo, Irene Arias, Computational evaluation of the flexoelectric effect in dielectric solids, J. Appl. Phys. 116 (2014) 093502.
- [9] R. Pelrine, R. Kornbluh, J. Joseph, Electrostriction of polymer dielectrics with compliant electrodes as a means of actuation, Sensors Actuators A 64 (1) (1998) 77–85.
- [10] R. Pelrine, R. Kornbluh, Q. Pei, J. Joseph, High-speed electrically actuated elastomers with strain greater than 100%, Science 287 (5454) (2000) 836–839.
- [11] R. Pelrine, R. Kornbluh, Q. Pei, S. Stanford, S. Oh, J. Eckerle, R.J. Full, M.A. Rosenthal, K. Meijer, Dielectric elastomer artificial muscle actuators: toward biomimetic motion, in: Smart Structures and Materials 2002: Electroactive Polymer Actuators and Devices (EAPAD), Vol. 4695, International Society for Optics and Photonics, SPIE, 2002, pp. 126–137.
- [12] G. Kofod, P. Sommer-Larsen, R. Kornbluh, R. Pelrine, Actuation response of polyacrylate dielectric elastomers, J. Intell. Mater. Syst. Struct. 14 (12) (2003) 787–793.
- [13] T. Li, C. Keplinger, R. Baumgartner, S. Bauer, W. Yang, Z. Suo, Giant voltage-induced deformation in dielectric elastomers near the verge of snap-through instability, J. Mech. Phys. Solids 61 (2) (2013) 611–628.
- [14] S. Rudykh, K. Bhattacharya, Gal deBotton, Snap-through actuation of thick-wall electroactive balloons, Int. J. Non-Linear Mech. 47 (2012) 206–209.
- [15] D.K. Vu, P. Steinmann, G. Possart, Numerical modelling of non-linear electroelasticity, Internat. J. Numer. Methods Engrg. 70 (6) (2007) 685–704.
- [16] A. Dorfmann, R.W. Ogden, Nonlinear electroelasticity, Acta Mech. 174 (3-4) (2005) 167-183.
- [17] A. Dorfmann, R.W. Ogden, Nonlinear electroelastic deformations, J. Elasticity 82 (2) (2006) 99-127.
- [18] A. Dorfmann, R.W. Ogden, Electroelastic waves in a finitely deformed electroactive material, IMA J. Appl. Math. 75 (4) (2010) 603–636.
- [19] D.K. Vu, P. Steinmann, A 2-D coupled BEM-FEM simulation of electro-elastostatics at large strain, Comput. Methods Appl. Mech. Engrg. 199 (17–20) (2010) 1124–1133.
- [20] D.K. Vu, P. Steinmann, On 3-D coupled BEM-FEM simulation of nonlinear electro-elastostatics, Comput. Methods Appl. Mech. Engrg. 201–204 (2012) 82–90.
- [21] R. Bustamante, A. Dorfmann, R.W. Ogden, On electric body forces and maxwell stresses in nonlinearly electroelastic solids, Internat. J. Engrg. Sci. 47 (11–12) (2009) 1131–1141.
- [22] R. Bustamante, J. Merodio, Constitutive structure in coupled non-linear electro-elasticity: Invariant descriptions and constitutive restrictions, Int. J. Non-Linear Mech. 46 (10) (2011) 1315–1323.
- [23] M. Itskov, V.N. Khiêm, A polyconvex anisotropic free energy function for electro- and magneto-rheological elastomers, Math. Mech. Solids (2014) 1–12.
- [24] J.E. Marsden, T.J.R. Hughes, Mathematical Foundations of Elasticity, Dover Civil and Mechanical Engineering, 1994.
- [25] J.M. Ball, Convexity conditions and existence theorems in nonlinear elasticity, Arch. Ration. Mech. Anal. 63 (4) (1976) 337-403.
- [26] A.J. Gil, R. Ortigosa, A new framework for large strain electromechanics based on convex multi-variable strain energies: variational formulation and material characterisation, Comput. Methods Appl. Mech. Engrg. 302 (2016) 293–328.
- [27] R. Ortigosa, A.J. Gil, A new framework for large strain electromechanics based on convex multi-variable strain energies: Finite element discretisation and computational implementation, Comput. Methods Appl. Mech. Engrg. 302 (2016) 329–360.
- [28] R. Ortigosa, A.J. Gil, A new framework for large strain electromechanics based on convex multi-variable strain energies: Conservation laws, hyperbolicity and extension to electro-magneto-mechanics, Comput. Methods Appl. Mech. Engrg. 309 (2016) 202–242.
- [29] R. Ortigosa, A.J. Gil, C.H. Lee, A computational framework for large strain nearly and truly incompressible electromecahnics based on convex multi-variable strain energies, Comput. Methods Appl. Mech. Engrg. 310 (2016) 297–334.
- [30] J.M. Ball, Energy-minimising configurations in nonlinear elasticity, Arch. Ration. Mech. Anal. 63 (4) (1976) 337-403.
- [31] J.M. Ball, Geometry, Mechanics and Dynamics, Springer, 2002, pp. 3-59.
- [32] J.M. Ball, F. Murat, W^{1,p}-quasiconvexity and variational problems for multiple integrals, J. Funct. Anal. 58 (3) (1984) 225–253.
- [33] J. Schröder, P. Neff, Invariant formulation of hyperelastic transverse isotropy based on polyconvex free energy functions, Int. J. Solids Struct. 40 (2003) 401–445.
- [34] J. Schröder, P. Neff, D. Balzani, A variational approach for materially stable anisotropic hyperelasticity, Int. J. Solids Struct. 42 (2005) 4352–4371.
- [35] J. Schröder, P. Neff, V. Ebbing, Anisotropic polyconvex energies on the basis of crystallographic motived structural tensors, J. Mech. Phys. Solids 56 (2008) 3486–3506.
- [36] J. Schröder, Anisotropic polyconvex energies, in: J. Schröder, P. Neff (Eds.), Poly-, Quasi- and Rank-One Convexity in Applied Mechanics, Volume 516 of CISM Courses and Lectures, in: CISM Courses and Lectures, vol. 516, Springer-Verlag, 2010, pp. 53–105.
- [37] J. Bonet, A.J. Gil, R. Ortigosa, On a tensor cross product based formulation of large strain solid mechanics, Int. J. Solids Struct. 84 (2016) 49-63.
- [38] J. Bonet, A.J. Gil, R. Ortigosa, A computational framework for polyconvex large strain elasticity, Comput. Methods Appl. Mech. Engrg. 283 (2015) 1061–1094.

- [39] A.C. Eringn, Microcontinuum Field Theories: Foundations and Solids, Springer, 1999.
- [40] J. Yvonnet, L.P. Liu, A numerical framework for modeling flexoelectricity and maxwell stress in soft dielectrics at finite strains, Comput. Methods Appl. Mech. Engrg. 313 (2017) 450–482.
- [41] D. Codony, P. Gupta, O. Marco, I. Arias, Modeling flexoelectricity in soft dielectrics at finite deformation, J. Mech. Phys. Solids 146 (2021) 104182.
- [42] Sheng Mao, Prashant K. Purohit, Nikolaos Aravas, Mixed finite-element formulations in piezoelectricity and flexoelectricity, Proc. R. Soc. Lond. Ser. A Math. Phys. Eng. Sci. 472 (2190) (2016) 20150879.
- [43] Roman Poya, Antonio J. Gil, Rogelio Ortigosa, Roberto Palma, On a family of numerical models for couple stress based flexoelectricity for continua and beams, J. Mech. Phys. Solids 125 (2019) 613–652.
- [44] A.T. McBride, D. Davydov, P. Steinmann, Modelling the flexoelectric effect in solids: A micromorphic approach, Comput. Methods Appl. Mech. Engrg. 371 (2020) 113320.
- [45] M.P. Bendsøe, O. Sigmund, Topology Optimization, Springer-Verlag, Berlin, 2003, p. xiv+370, Theory, methods and applications.
- [46] M. Wang, X. Wang, D. Guo, A level-set method for structural topology optimization, Comput. Methods Appl. Mech. Engrg. 192 (2003) 227–246.
- [47] G. Allaire, F. Jouve, Toader. A., Structural optimization using sensitivity analysis and a level-set method, J. Comput. Phys. 194 (2004) 363–393.
- [48] M. Burger, R. Stainko, Phase-field relaxation of topology optimization with local stress constraints, SIAM J. Control Optim. 192 (2003) 147–1466.
- [49] J. Sokolowski, A. Zochowski, On the topological derivative in shape optimization, SIAM J. Control Optim. 37 (1999) 1251–1272.
- [50] D.J. Munk, G.P. Steven, Topology and shape optimization methods using evolutionary algorithms: a review, Struct. Multidiscip. Optim. 52 (2015) 613–631.
- [51] Z. Kang, X. Wang, Topology optimization of bending actuators with multilayer piezoelectric material, Smart Mater. Struct. 19 (7) (2010) 075018.
- [52] X. Zhang, A. Takezawa, Z. Kang, Topology optimization of piezoelectric smart structures for minimum energy consumption under active control, Struct. Multidiscip. Optim. 58 (2018) 185–199.
- [53] M. Kögl, E. Silva, Topology optimization of smart structures: Design of piezoelectric plate and shell actuators, Smart Mater. Struct. 14 (2005) 387–399.
- [54] A. Donoso, J. Bellido, Systematic design of distributed piezoelectric modal sensors/actuators for rectangular plates by optimizing the polarization profile, Struct. Multidiscip. Optim. 38 (2009) 347–356.
- [55] A. Donoso, J. Bellido, J. Chacón, Numerical and analytical method for the design of piezoelectric modal sensors/actuators for shell-type structures, Internat. J. Numer. Methods Engrg. 81 (2009) 1700–1712.
- [56] J.L. Sánchez-Rojas, J Hernando, A Donoso, J. Bellido, T. Manzaneque, A. Ababneh, H. Seidel, U. Schmid, Modal optimization and filtering in piezoelectric microplate resonators, J. Micromech. Microeng. 20 (2010) 055027.
- [57] D. Ruiz, J. Bellido, A. Donoso, J.L. Sanchez-Rojas, Design of in-plane piezoelectric sensors for static response by simultaneously optimizing the host structure and the electrode profile, Struct. Multidiscip. Optim. 48 (2013) 1023–1026.
- [58] D. Ruiz, J. Bellido, A. Donoso, Design of piezoelectric modal filters by simultaneously optimizing the structure layout and the electrode profile, Struct. Multidiscip. Optim. 53 (2015) 715–730.
- [59] D. Ruiz, Alex Díaz-Molina, O. Sigmund, A. Donoso, J. Bellido, J.L. Sánchez-Rojas, Optimal design of robust piezoelectric unimorph microgrippers, Appl. Math. Model. 55 (2018) 1–12.
- [60] D. Ruiz, O. Sigmund, Optimal design of robust piezoelectric microgrippers undergoing large displacements, Struct. Multidiscip. Optim. 55 (2018) 71–82.
- [61] P.H. Nakasone, E.C.N. Silva, Dynamic design of piezoelectric laminated sensors and actuators using topology optimization, J. Intell. Mater. Syst. Struct. 21 (16) (2010) 1627–1652.
- [62] E. Padoin, I.F. Santos, Perondi E.A., Topology optimization of piezoelectric macro-fiber composite patches on laminated plates for vibration suppression, Struct. Multidiscip. Optim. 59 (2019) 941–957.
- [63] E.C.N. Silva, J.S. Fonseca, F.M de Espinosa, Design of piezocomposite materials and piezoelectric transducers using topology optimization: Part I, Arch. Comput. Methods Eng. 6 (1999) 117–182.
- [64] C. Lundgaard, O. Sigmund, A density-based topology optimization methodology for thermoelectric energy conversion problems, Struct. Multidiscip. Optim. 57 (2018) 1427–1442.
- [65] C. Lundgaard, O. Sigmund, Design of segmented off-diagonal thermoelectric generators using topology optimization, Appl. Energy 236 (2019) 950–960.
- [66] C. Lundgaard, O. Sigmund, Design of segmented thermoelectric peltier coolers by topology optimization, Appl. Energy 239 (2019) 1003–1013.
- [67] N. Wang, H. Guo, B. Chen, X. Zhang, Design of a rotary dielectric elastomer actuator using topology optimization method, in: 2017 International Conference on Manipulation, Automation and Robotics At Small Scales (MARSS), 2017, pp. 1–6.
- [68] E. Bortot, O. Amir, G. Shmuel, Topology optimization of dielectric elastomers for wide tunable band gaps, Int. J. Solids Struct. 143 (2018) 262–273.
- [69] P.I. Galich, S. Rudykh, Shear wave propagation and band gaps in finitely deformed dielectric elastomer laminates: Long wave estimates and exact solution, J. Appl. Mech. 84 (2017).
- [70] R. Getz, G. Shmuel, Band gap tunability in deformable dielectric composite plates, Int. J. Solids Struct. 128 (2017) 11-22.
- [71] R. Getz, D.M. Kochmann, G. Shmuel, Voltage-controlled complete stopbands in two-dimensional soft dielectrics, Int. J. Solids Struct. 113–114 (2017) 24–36.

- [72] R. Ortigosa, J. Martínez-Frutos, D. Ruiz, A. Donoso, J. C. Bellido, Density-based topology optimisation considering nonlinear electromechanics, Struct. Multidiscip. Optim., in print.
- [73] J. Martínez-Frutos, R. Ortigosa, A.J. Gil, In-silico design of electrode meso-architecture for shape morphing dielectric elastomers, 157, 2021, 104594,
- [74] R. Ortigosa, Martínez-Frutos. J., Constructive and destructive interplay between piezoelectricity and flexoelectricity in flexural sensors and actuators, Struct. Multidiscip. Optim. 64 (2021) 3681–3703.
- [75] R.M. McMeeking, C.M. Landis, Electrostatic forces and stored energy for deformable dielectric materials, J. Appl. Mech. 72 (4) (2008) 581–590.
- [76] C. Miehe, D. Vallicotti, D. Zäh, Computational structural and material stability analysis in finite electro-elasto-statics of electro-active materials, Internat. J. Numer. Methods Engrg. 102 (10) (2015) 1605–1637.
- [77] S.S Nanthakumar, X. Zhuang, H.S Park, T. Rabczuk, Topology optimization of flexoelectric structures, J. Mech. Phys. Solids 105 (2017) 217–234.
- [78] H. Ghasemi, H.S. Park, T. Rabczuk, A multi-material level set-based topology optimization of flexoelectric composites, Comput. Methods Appl. Mech. Engrg. 332 (2018) 47–62.
- [79] E. Holmberg, B. Torstenfelt, A. Klarbring, Stress constrained topology optimization, Struct. Multidiscip. Optim. 48 (2013) 33-47.
- [80] K.J. Bathe, Finite Element Procedures, Prentice Hall, 1996.
- [81] J. Bonet, A.J. Gil, R.D. Wood, Nonlinear Continuum Mechanics for Finite Element Analysis: Statics, Cambridge University Press, 2016.
- [82] E.A. de Souza Neto, D. Períc, D.R.J. Owen, Computational Methods for Plasticity. Theory and Applications, John Wiley and Sons Ltd, 2008.
- [83] O. González, A.M. Stuart, A First Course in Continuum Mechanics, Cambridge University Press, 2008.
- [84] S. Skatulla, C. Sansour, A. Arockiarajan, A multiplicative approach for nonlinear electro-elasticity, Comput. Methods Appl. Mech. Engrg. 245–246 (2012) 243–255.
- [85] T.E. Bruns, D.A. Tortorelli, Topology optimization of non-linear elastic structures and compliant mechanisms, Comput. Methods Appl. Mech. Engrg. 190 (26) (2001) 3443–3459.
- [86] Bourdin B., Filters in topology optimization, Internat. J. Numer. Methods Engrg. 50 (9) (2001) 2143–2158.
- [87] F. Wang, B.S. Lazarov, O. Sigmund, On projection methods, convergence and robust formulations in topology optimization, Struct. Multidiscip. Optim. 43 (6) (2011) 767–784.
- [88] Bin Chen, Matthias Kollosche, Mark Stewart, James Busfield, Federico Carpi, Electrical breakdown of dielectric elastomers: influence of compression, electrode's curvature and environmental humidity, in: Yoseph Bar-Cohen, Frédéric Vidal (Eds.), Electroactive Polymer Actuators and Devices (EAPAD) 2016, Vol. 9798, International Society for Optics and Photonics, SPIE, 2016, pp. 76–85.
- [89] F. Brezzi, M. Fortin, Mixed and Hybrid Finite Element Methods, Springer-Verlag New York, Inc., New York, NY, USA, 1991.
- [90] D. Chapelle, K.J. Bathe, The inf-sup test, Comput. Struct. 47 (1993) 537-545.
- [91] O. Sigmund, Manufacturing tolerant topology optimization, Acta Mech. Sinica 25 (2009) 227-239.
- [92] J. Martínez-Frutos, F. Kessler, Robust optimal shape design for an elliptic PDE with uncertainty in its input data, ESAIM Control Optim. Calc. Var. 21 (2015).
- [93] C. Svanberg, The method of moving asymptotes: a new method for structural optimization, Appl. Energy 24 (1987) 359-373.
- [94] K. Liu, A. Tovar, An efficient 3D topology optimization code written in matlab, Struct. Multidiscip. Optim. 50 (2014) 1175-1196.